

#### The Impact of Organic Fertilizer Among Smallholder Farmers in Sub-Saharan Africa

Jonathan Lehe, Abraham Salomon, Vaishnavi Surendra, Jessica Zhu

This study is produced by Precision Development (precisiondev.org) and funded by Livelihoods Impact Fund. The authors thank Ellie Chen and Kamran Khan Niazi for excellent research assistance, and Tomoko Harigaya, Supriya Ramanathan, Catherine Whetstone, and Ani Vallabhaneni for helpful comments.

> The authors may be contacted at: Lehe: jlehe@precisiondev.org Salomon: asalomon@precisiondev.org Surendra: vsurendra@precisiondev.org Zhu: jzhu@precisiondev.org

> > USUAL DISCLAIMERS APPLY

**JANUARY 20, 2025** 





## Executive Summary

The agriculture sector is a key contributor to Africa's economy, but is held back by large yield gaps. These yield gaps have persisted in Africa despite dramatic increases in fertilizer application over the last two decades, driven primarily by the use of imported inorganic fertilizers. There is little evidence that continuing to increase the application of imported inorganic fertilizers alone will generate the desired outcome of increasing yields. At the same time, the increased use of imported inorganic fertilizer has resulted in negative soil-health, environmental, and economic outcomes in Africa.

Application of organic fertilizer, in balance with inorganic fertilizer, is a promising prospective solution to these problems. A renewed call to "triple domestic production and distribution of certified quality organic and inorganic fertilizers by 2034" was recently announced in the Nairobi Declaration of the Africa Fertilizer and Soil Health Summit held May 7–9, 2024 in Nairobi. This declaration includes an increased focus on domestic fertilizer production, the need for a balanced approach between increased productivity and soil health restoration, and the efficient use of organic and inorganic fertilizers, along with other complementary inputs (African Union, 2024).

This study examines the relationship between organic fertilizer usage and its various benefits in the setting of smallholder farmers in sub-Saharan Africa. There is robust agronomic evidence of the impact of organic fertilizer on soil fertility, and agricultural productivity and profitability, and suggestive evidence of the environmental benefits and economic viability of organic fertilizer. Additional rigorous evaluations on organic fertilizers' impacts and cost-effectiveness could be helpful to inform policy recommendations further. The adoption of organic fertilizer remains low because of insufficient local manufacturing capacity, limited knowledge of the benefits of using organic fertilizers, and labor and financial constraints faced by farmers. Commercial organic fertilizers coupled with digital extension services have the potential to overcome some of those adoption barriers, and therefore improve farmers' usages and welfare.



#### TABLE OF CONTENTS

| 1. Introduction  | 3  |  |  |  |
|--|----|--|--|--|
| 1.1 Yield gaps   | 3  |  |  |  |
| 1.2 Soil fertility   | 4  |  |  |  |
| 1.3 The role of fertilizer in agricultural productivity                        | 5  |  |  |  |
| 1.3.1 Inorganic fertilizer   | 5  |  |  |  |
| 1.3.2 Organic fertilizer   | 6  |  |  |  |
| 1.4. Outline of the paper  | 7  |  |  |  |
| 2. The impact of organic fertilizer on agronomic outcomes                      |    |  |  |  |
| 2.1 The impact of organic fertilizers on soil fertility                        | 8  |  |  |  |
| 2.2 The impact of organic fertilizer on crops: yields, nutrient content, and   |    |  |  |  |
| efficiency of fertilizers  | 9  |  |  |  |
| 3. Heterogeneity in the impact of organic fertilizer on agronomic outcomes     | 12 |  |  |  |
| 3.1 Agroecological conditions  | 12 |  |  |  |
| 3.2 Crops and growing patterns   | 13 |  |  |  |
| 3.3 Application rates  | 14 |  |  |  |
| 3.4 Organic fertilizer type and quality  | 14 |  |  |  |
| 4. The impact of organic fertilizer on environmental outcomes                  | 16 |  |  |  |
| 4.1 Impacts due to reduction in inorganic fertilizer use                       | 16 |  |  |  |
| 4.2 Impacts due to organic fertilizer use                                      | 17 |  |  |  |
| 4.3 Impacts due to production of organic fertilizers                           | 17 |  |  |  |
| 5. The impact of organic fertilizer on economic outcomes                       |    |  |  |  |
| 5.1. Economic impact in agronomic trials                                       | 19 |  |  |  |
| 5.2. Economic impact in a real-world setting                                   | 20 |  |  |  |
| 5.3. Factors contributing to the differences in productivity between agronomic |    |  |  |  |
| trials and real-world settings   | 20 |  |  |  |
| 6. Cost-effectiveness of organic fertilizer                                    | 22 |  |  |  |
| 7. Adoption of organic fertilizer  | 24 |  |  |  |
| 8. Macro perspective of commercial organic fertilizer sector                   | 26 |  |  |  |
| 9. Conclusion  | 27 |  |  |  |
| 10. Figure and Tables  | 28 |  |  |  |
| Reference  | 30 |  |  |  |
| Appendix A: Soil fertility and soil organic matter                             | 40 |  |  |  |
| A.1 Definition of soil fertility and soil organic matter                       | 40 |  |  |  |
| A.2. Methods that increase soil organic matter                                 | 40 |  |  |  |
| Appendix B: A case study of a specific commercial organic fertilizer           | 42 |  |  |  |
| B.1 Introduction   | 42 |  |  |  |
| B.2 Farmer awareness, adoption, and perceptions of Evergrow                    | 42 |  |  |  |
| B.3 Impact of Evergrow on yields and revenues                                  | 43 |  |  |  |
| B.4 Conclusion   | 45 |  |  |  |

## 1. Introduction

#### Key takeaways:

- Yield gaps persist in Kenya and the region and are correlated with poverty.
- Soil degradation is one of the key challenges in closing the yield gap.
- Many approaches can be used to improve yields: adequate fertilization is an important element to yield improvement.
- Historically policy has focused on inorganic fertilizer. But inorganic fertilizer alone does not counteract long term soil fertility decline.
- Introducing organic fertilizer contributes to improving yields and soil fertility; but there are challenges with homemade organic fertilizer. Commercial organic fertilizer industry is nascent in the LMIC settings.
- This paper assesses the available evidence on the agronomic and economic impact of organic fertilizers and discusses the opportunities and challenges to substantially increase the adoption at scale.

The agriculture sector is the backbone of sub-Saharan Africa's economy. It employs around 60 percent of the labor force, produces approximately 22 percent of the gross domestic product (GDP), and accounts for more than 10 percent of export revenues (Wudil et al., 2022). In Kenya, the agriculture sector employs more than 40 percent of the total population and 70 percent of the rural population, and contributes 33 percent of GDP as the largest contributing sector (U.S. Agency for International Development, 2023).

#### 1.1 Yield gaps

Despite the importance of the agriculture sector in sub-Saharan Africa and in Kenya specifically, many farming systems in the region are characterized by large, persistent yield gaps and stagnating agricultural growth rates (Food and Agriculture Organization [FAO], 2024; Kenya National Bureau of Statistics [KNBS], 2023). A yield gap is the difference between measured or reported actual yields, and the potential yield a crop could achieve in a particular environment (van Ittersum et al., 2013). Though there are various methods of calculating this gap with differing levels of modeling and assumptions, results consistently show a remarkably large yield gap for Kenya and many other countries in the region, with average yields in the range of 70-80 percent of the potential yield for maize, 60-80 percent for beans and sorghum, and 40-60 percent for rice (Grassini et al., 2017; Global Yield Gap Atlas, 2024).

More importantly, agricultural productivity has stagnated in recent years in Kenya, as in many other areas in sub-Saharan Africa. In Kenya, the real agricultural growth rate averaged only 2.4 percent from 2018 to 2022, including negative growth in two of these five years (KNBS, 2023). As a whole, the average annual growth rates of Africa's agricultural value added slowed from 5.4 percent per year in 2002-2011 to 3.5 percent in 2012-2021 (Food and Agriculture Organization [FAO], 2023). The sluggish growth rates make closing the yield gaps even more difficult.



In large part due to these yield gaps, incomes of smallholder farmers in sub-Saharan Africa remain stubbornly low. For example, the median monthly income for smallholder farmers is only USD 16 in Zambia, USD 19 in Tanzania, and USD 28 in Kenya (Karlyn, 2017). These yield gaps also lead to high levels of undernourishment in Africa, with nearly 20 percent of the population undernourished as of 2014 (Giller, 2020). Nearly 70 percent of households in Africa depend on agriculture for part of their livelihoods, and food regularly makes up more than 50 percent of the expenses of many households in Africa (van de Ven et al., 2021), making yield improvements critical to solving nutrition and food security challenges across the continent (Giller, 2020). These concerning trends of yield gaps and stagnated growth demand new innovations and solutions.

#### 1.2 Soil fertility

Several challenges have contributed to the persistent yield gaps, including land fragmentation, soil fertility decline, dated management practices, climate change, and institution- and market-related factors (Birch, 2018; D'Alessandro et al., 2015). Of these challenges, the decline in soil fertility is one of the most significant.<sup>1</sup> Degradation of soil fertility across sub-Saharan Africa is well documented, particularly in terms of soil organic matter (SOM) and nutrient balances, as well as other indicators of soil productivity (Nandwa & Bekunda, 1998). The correlation between the decline in soil fertility and the decline in crop yields is demonstrated in multiple studies (Bekunda et al., 2005; Moebius-Clune et al., 2011; Mutuku et al., 2020; Tittonell & Giller, 2013; Tully et al., 2015). These studies relate soil fertility decline to various parameters, notably SOM or soil organic carbon (SOC), soil nutrient status, soil acidity, and soil physical features.

Farming itself is the major cause of these declines in soil fertility, as many basic agricultural practices have been shown to contribute to soil fertility decline. These practices include disturbing the soil, exporting soil nutrients (in crops), and reducing permanent soil cover (Kimetu et al., 2008, Means et al., 2022). In a meta-analysis of 102 studies, Kopittke et al. (2017) show that converting native vegetation to cropping land results in the depletion of critical soil nutrients, including decreases of 43 percent of carbon (C,) 42 percent of nitrogen (N), 27 percent of phosphorus (P), and 33 percent of sulfur (S). Furthermore, when land is converted to farmland, it becomes more susceptible to erosion, and the increasing levels of soil erosion also contribute to the decline of soil fertility (Kogo et al., 2020).

Continuous agricultural cultivation without adequate replacement of nutrients and organic matter causes negative nutrient balances in soils, and reduces soil fertility through loss of SOM (Moebius-Clune et al., 2011; Nyberg et al., 2012; Tully et al., 2015). Crops extract nutrients from soil, and if more nutrients are extracted than are replaced by natural processes or human intervention, this negative nutrient balance depletes the soil fertility over time, as has been observed across several studies of African farming systems (Chianu et al., 2012; Vanlauwe & Giller, 2006).

Improving the nutrient balance can be achieved by increasing input of nutrients, which has an immediate effect on crop growth and yield. However, improving SOM and the physical properties of soils requires not only nutrient inputs, but management of organic resources, including by external addition of organic amendments (Kihara et al., 2016; Roobroeck et al., 2021). SOM (or the closely related SOC) can be defined as organic materials found in soils; these materials originate from living organisms, especially plants and microbes (Chenu et al., 2024). The level of SOM in soils is an essen-

1

Appendix A provides a detailed discussion on soil fertility and soil organic matter.



tial component of soil fertility, and it contributes to crop productivity (Lal, 2006; Manlay, Feller & Swift, 2007). Many studies across sub-Saharan Africa and the rest of the world show the strong relationship between crop yields and SOM (Lal, 2006). Both nutrient management and organic matter management practices offer significant potential for improvement in soil fertility (Gicheru, 2012; Tittonell & Giller, 2013).

#### 1.3 The role of fertilizer in agricultural productivity

One generally accepted policy recommendation to close the yield gap has been to increase fertilizer application. For example, in the Abuja Declaration on Fertilizer for the Africa Green Revolution, announced in June 2006, the African Union member states resolved to increase fertilizer use from 8 kilograms per hectare (kg/Ha) in 2006 to an average of at least 50 kg/Ha by 2015, including both inorganic and organic fertilizers (African Union, 2006).

#### 1.3.1 Inorganic fertilizer

Inorganic fertilizer is defined as "a nutrient-rich fertilizer produced industrially by chemical processes or mineral extraction" by the Food and Agriculture Organization (FAO) Committee on Agriculture and has various types, such as synthetic, mineral, or chemical fertilizers (Food and Agriculture Organization [FAO], 2018). This paper uses the generalized term "inorganic fertilizer" and does not differentiate among those specific types (including urea).

On one hand, various studies have shown that the application of inorganic fertilizer is an important factor in maintaining and increasing crop yields (Giller, 2020; Lassaletta et al., 2014; Stewart et al., 2005). However, on the other hand, despite the near tripling of fertilizer application rates in Africa since the Abuja Declaration, to 22 kg/Ha in 2022 (Malpass, 2022), a proportional decrease in the yield gap has not been observed (Dimkpa et al., 2023; Jayne et al., 2018). For example, in countries such as Kenya, where fertilizer application rates have achieved the Abuja Declaration targets of 50 kg/ Ha, yields have continued to stagnate (Ng'ang'a, 2024). The application of inorganic fertilizer without adequate management of organic amendments could result in soil fertility decline that could lead to reduced yields in the long term, especially when inorganic fertilizer applications are interrupted (Tittonell & Giller, 2013). There is little evidence suggesting that continuing to increase the application of inorganic fertilizers alone will generate the desired outcome of increasing yields.

Moreover, the dramatic increase in fertilizer use in Kenya and many other parts of Africa has been driven primarily by an increase in imported inorganic fertilizers (Ariga & Jayne, 2011), with less than one percent of fertilizer use coming from commercial organic fertilizers in Kenya in 2022 (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b). According to this business report, the heavy reliance on inorganic fertilizer compared to organic fertilizer, and on imported fertilizer compared to locally produced fertilizer, has contributed to several additional problems, including soil degradation through acidification and nutrient mining; increased greenhouse gas emissions from fertilizer transportation; and macroeconomic shocks due to the rising price of fertilizer imports, especially following Russia's invasion of Ukraine and the subsequent global food and fertilizer crisis (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023a).



#### 1.3.2 Organic fertilizer

By contrast, organic fertilizers are derived from organic materials, primarily plant residues and livestock manures. The specific use of the term organic fertilizer lacks consistency (Vanlauwe and Giller, 2006). The International Standards Organization defines organic fertilizer broadly as "material containing carbon or one or more elements other than hydrogen and oxygen mainly of plant and/or animal origin added either directly to the plant or to the soil" (ISO, 2022). National standards for the composition of organic fertilizers are in place to regulate commercial organic fertilizers in the formal market, but do not necessarily extend to informal, farm-sourced organic fertilizers. For example, the Kenya KS 2290: 2023 organic fertilizer standard includes compositional requirements such as C:N ratio <20, moisture content <40 percent, N content >1 percent, organic matter content >20 percent, and the total combined N, P, and potassium (K) content >3.5 percent <sup>2</sup> (Kenya Bureau of Standards [KEBS], 2023). These standards also include labeling requirements, and maximum levels of contaminants and pathogens. For ease of comparison with commercial organic fertilizers, when referring to non-commercial organic fertilizer, this paper focuses on composted material and manure, and excludes the discussion on uncomposted plant residues (e.g., green manures, crop residue, cover crops) and other soil conditioners (e.g., biochar).

Organic fertilizer not only contributes to increases in crop yields in the short term, but also contributes to soil fertility improvements in the long term by increasing SOM, which improves many soil physical and chemical properties (Kopittke et al., 2020; Lal, 2006). These improvements in soil fertility lead to further impacts on crop yields in the long term, and also increase certain nutritional parameters of crops.

In the short term, the usage of organic fertilizer leads to direct improvements in yield. This is due to direct increases in nutrient supply to the crops as well as from improved nutrient availability from soil, after organic fertilizer application. These higher yields contribute directly to an improvement in farmer incomes. The usage of organic fertilizer also often reduces the cost of production by improving efficiency of inorganic fertilizers when used in combination. The application of organic fertilizer allows farmers to apply less inorganic fertilizer and get higher yields per unit of both organic and inorganic fertilizers when they are applied together. These two effects of higher yields and lower costs contribute to farmers achieving higher profits. Moreover, the improved efficiency of input usage, particularly in the use of inorganic fertilizer, contributes to reduced negative environmental impacts. These include reduction in the potential for nutrient leaching and gaseous losses, which can harm ecosystems and human health.

In the long term, the usage of organic fertilizers increases the levels of SOM. The level of SOM affects many soil features, including soil structure, cation exchange capacity, available water-holding capacity, pH level, organo-mineral complex formation, and microbial diversity (Adekiya et al., 2020; Franzluebbers, 2002; Kimetu et al., 2008; Kopittke et al., 2020; Lal, 2006; Manlay et al., 2007; Nelson & Su, 2010). Thus, the improvement in SOM levels creates a positive feedback loop for producing higher agricultural yields. It is worth noting that changes in SOM status are long-term outcomes, often the result of repeated applications of organic fertilizer over time.

Combining organic and inorganic fertilizer is one of the key features of the Integrated Soil Fertility Management (ISFM) approach. It has been shown to result in improved soil health, agricultural yields, and input efficiency of both types of fertilizers by a large set of scientific literature (Vanlauwe, 2003; Vanlauwe & Giller, 2006). Table 1 provides a brief summary of the characteristics of inorganic and organic fertilizer.



#### 1.4. Outline of the paper

This paper aims to present detailed evidence of the relationship between the application of organic fertilizer and its agronomic, economic, and environmental benefits. Increasing the application of organic fertilizers is expected to have multiple positive impacts for smallholder farmers in both the short and long term. This paper shows the agronomic impact of organic fertilizers without differentiating whether they are commercially produced or not, as these share the same objectives and impact mechanisms, while discussing the economic benefits and adoption barriers of commercial and non-commercial organic fertilizer separately. The specific impact mechanisms are summarized in Figure 1.

The remainder of this paper is organized as follows. Section 2 describes the empirical evidence for how organic fertilizers improve soil fertility, yields, and other agronomic outcomes. Section 3 describes the heterogeneity in this evidence. Section 4 describes the evidence for the impact of organic fertilizers on environmental outcomes. Section 5 describes the evidence for the impact of organic fertilizers on farmer revenue, profitability, and other economic outcomes, in agronomic trials and real-world settings. Section 6 discusses the cost effectiveness of organic fertilizers. Section 7 provides details on the current adoption of organic fertilizers in sub-Saharan Africa. Section 8 briefly discusses additional macro perspectives of the growth of the commercial organic fertilizer sector. Section 9 concludes. Appendix A provides a detailed discussion on soil fertility and soil organic matter. A case study of one specific commercial organic fertilizer – Evergrow – is presented in Appendix B.



## 2. The impact of organic fertilizer on agronomic outcomes

#### **Key takeaways:**

- Organic fertilizers improve soil structure and physical properties.
- Organic fertilizers increase soil nutrient retention and availability.
- Organic fertilizers may promote soil biological activity.
- Organic fertilizers can improve the efficiency and responsiveness of crops to inorganic fertilizers.

#### 2.1 The impact of organic fertilizers on soil fertility

Soil degradation influences soil fertility and agricultural productivity. There is robust evidence from agronomic trials across multiple settings showing that the use of organic fertilizers mitigates soil degradation by improving soil quality through various mechanisms, which in turn boosts yields. Organic fertilizers: enhance the physical structure and biochemical activity of soil, leading to the gradual decomposition of SOM and long-term improvements in soil quality and fertility (Li et al., 2017; Oyetunde-Usman et al., 2021); improve soil water content (Bhanwaria et al., 2022; Lin et al., 2023); augment soil enzyme function (Lazcano et al., 2021); increase water and nutrient retention in the soil (Wang et al., 2020); and stimulate the transformation of soil nutrients and improve the accessibility of nutrients within the soil (Ahmad et al., 2016). Apart from these direct benefits, the improvement in soil's physical and biological properties also improves soil's responsiveness to inorganic fertilizers (Kearney et al., 2012; Marenya & Barrett, 2009b), and could encourage the use of inorganic fertilizers when market conditions permit (Marenya & Barrett, 2009a).

The use of organic fertilizers improves N, P and K levels in the soil. In their meta-analysis, Fan et al. (2023) find that, relative to chemical fertilizers, organic fertilizers increase total N by 32.79 percent, total P by 23.97 percent, total K by 44.91 percent, available P by 14.46 percent, and available K by 16.21 percent. The use of organic fertilizers has also been shown to improve plant P nutrition in tropical soils, not only by affecting the physicochemical availability of P but also by stimulating microbial activity, which in turn boosts the biological cycling of P (Kearney et al., 2012; Nest et al., 2014). The ability of organic fertilizer to elevate soil pH also decreases the adsorption capacity of elements like aluminum, thereby releasing P back into the soil for plant use.

Tovihoudji et al. (2018) report changes in soil properties after four years of maize cropping using 6 tonnes per hectare (t/ha) of manure, with or without inorganic N combinations, in Benin. They find a doubling of SOC content after four years of manure application, which increases with higher inorganic N applications. Synthetic N applications without manure increase SOC content slightly. The authors also find a very strong correlation between SOC and grain yields, with an R<sup>2</sup> of 0.71. Similar responses are seen in manure-treated soils for available P.



Organic fertilizers improve the physical properties of soil, and foster the growth of beneficial microorganisms (Lazcano et al., 2021; Leroy, 2008; Lori et al., 2017; Moeskops et al., 2012) by increasing SOM and nutrient availability (Lazcano et al., 2021). Organic fertilizers also increase the activity of microarthropods and larger organisms, such as earthworms and springtails (Forge et al., 2003; Leroy et al., 2008; Mäder et al., 2002; Pfotzer & Schüler, 1997). These organisms facilitate N release, promote soil aggregate stability, and suppress diseases. In a meta-analysis of 56 studies, Lori et al. (2017) find that, when compared with conventional farming systems, organic farming systems (rather than just the use of organic fertilizers) have between 32 percent and 84 percent greater microbial biomass C, microbial biomass N, and total phospholipid fatty acids, and dehydrogenase, urease and protease activities, which indicate higher microbial activity. In another meta-analysis of 124 studies evaluating the impact of organic and inorganic fertilizers on soil properties and tomato production, Fan et al. (2023) find that organic fertilizers increase soil bacteria by 5.94 percent, urease by 22.32 percent, and catalase by 17.68 percent, when compared with inorganic fertilizers. In addition, increasing organic C improves soil aggregation (Kay & Angers, 2001) and can reduce soil penetration resistance, thereby facilitating better root growth and water infiltration (Kay et al., 1997). Enhanced soil structure is also pivotal for improving water retention and reducing erosion (Lynch & Bragg, 1985; Tisdall & Oades, 1982).

The evidence above demonstrates the impact of organic fertilizers on soil fertility. However, when considering organic fertilizers versus inorganic fertilizers, it is important to note that organic fertilizers generally release nutrients slowly, thus improving the soil's nutrient-holding capacity and offering a sustained supply of nutrients over time. Inorganic fertilizers, on the other hand, can make nutrients immediately available, which is particularly necessary during the peak growth stages of crops. Thus combining organic fertilizer with inorganic fertilizer not only addresses the immediate nutrient demands of crops, but also contributes to the maintenance of SOM and the overall biological functioning of soils. There may be a trade off in organic fertilizer quality and the interaction effect of organic and inorganic fertilizers, with lower yield response to inorganic fertilizers when combined with higher quality organic fertilizers (Kimetu et al., 2008). Some studies cite higher quality organic fertilizers may contribute less to SOM formation, others show no influence of organic amendment quality, and others find effects of quality on specific pools of carbon in soils, but there is generally an imperfect understanding of how organic amendment guality affects SOM accumulation (Fonte et al., 2009; Kimetu et al., 2008; Sonsri et al, 2022). The strategic use of organic inputs alongside inorganic fertilizers can enhance nutrient cycling, mitigate nutrient losses, and correct minor nutrient imbalances, thereby alleviating crop growth limitations more effectively than either type of fertilizer used alone (Kearney et al., 2012).

## 2.2 The impact of organic fertilizer on crops: yields, nutrient content, and efficiency of fertilizers

In addition to the broad effects discussed above, organic fertilizers also affect crop yield and may improve crop quality (in terms of crop nutritional content). Evidence from agronomic trials conducted across sub-Saharan Africa demonstrates improvements in yield and crop quality following the use of organic fertilizer (Anyega et al., 2021; Beesigamukama, Mochoge, Korir & Fiaboe, 2020; Beesigamukama, Mochoge, Korir & Musyoka, 2020; Chivenge et al., 2021; Fan et al., 2023; Gram et al., 2020; Houben et al., 2020; Kearney et al., 2012; Nsoanya & Nweke, 2015; Vanlauwe et al., 2011). While some studies compare using organic fertilizer with not using fertilizer, others compare organic with



inorganic fertilizers, and still others compare using a combination of organic and inorganic fertilizers with using no fertilizer, or with using either organic or inorganic fertilizers. These findings are discussed below.

Fan et al. (2023) analyze data from 124 agronomic studies conducted across the world in their meta-analysis of the impact of organic fertilizers on tomato yield and quality, relative to control groups where only inorganic fertilizers are applied. Their analysis of the impact of organic fertilizers on soil properties and soil fertility is discussed in section 2.1. When it comes to tomato nutrient content, Fan et al. (2023) find that organic fertilizer reduces fruit nitrate by 12.66 percent, and increases fruit sugar by 4.79 percent, fruit lycopene by 4.80 percent, and fruit protein by 13.82 percent. In addition, they find that organic fertilizer increases the yield by 3.48 percent compared with a control group where only inorganic fertilizer was applied. In a study focusing on maize in Kenya, Beesigamukama, Mochoge, Korir & Fiaboe (2020) compare N (from urea) replacement values between two organic fertilizer (black soldier fly frass and a commercial organic fertilizer), each combined with inorganic P and K. They find that there were small increases in yields with increasing N-equivalent rates of commercial organic fertilizer (in one season out of two) but not frass, whereas N replacement equivalents tended to be higher for frass.

When it comes to plant uptake of N, Beesigamukama, Mochoge, Korir & Musyoka (2020) find that the release of N in soils treated with organic fertilizers was slow during early-stage maize growth, leading to insufficient N availability relative to the amount of N required by plants (N synchrony). This lack of N synchrony was less pronounced with frass than with the commercial organic fertilizer, but suggests that the N in inorganic fertilizer might be a necessary supplement to organic fertilizer. However, there is probably variability in this, based on the type of organic fertilizer and its nutrient content.

A number of agronomic trials that use a combination of organic and inorganic fertilizers find that this combination leads to better outcomes than either fertilizer type alone. In a study with cabbage in Uganda, Kearney et al. (2012) find that a combination of organic (with 7.1 percent organic matter and 0.34 percent N) and inorganic fertilizer produces 26 percent significantly higher biomass than inorganic fertilizer alone produces. Anyega et al. (2021) study the impact of different organic fertilizers (frass, spent grain, commercial organic fertilizer), inorganic fertilizers, their combination, and not using fertilizer on tomato, kale, and French beans in agronomic trials in both greenhouse and open-field conditions over two cropping seasons in Kenya. They find that vegetable yields achieved from a combination of organic fertilizer (frass) and inorganic fertilizer are significantly higher than from both the control condition and inorganic fertilizers alone. The combination gives both higher agronomic N-use efficiency (increase in yield per unit of N fertilizer applied) and vegetables with the highest crude protein and ash concentrations. This suggests that the combination of organic and inorganic fertilizers boosts the yield and nutritional quality of vegetable crops.

Multiple studies (Anyega et al., 2021; Beesigamukama, Mochoge, Korir & Fiaboe, 2020; Beesigamukama, Mochoge, Korir & Musyoka, 2020) and meta-analyses conducted with maize in sub-Saharan Africa (Chivenge et al., 2021; Gram et al., 2020; Vanlauwe et al., 2011), demonstrate that the combined use of organic fertilizers with inorganic fertilizers increases maize yields and the agronomic efficiency (increase in yield per unit of fertilizer nutrient applied) of applied nutrients, relative to inorganic fertilizers alone, and relative to organic fertilizers alone in certain cases.



Tovihoudji et al. (2018) find some evidence for higher N agronomic efficiency for manure without combination with inorganic N, though this changes significantly from year to year, and the relationships between organic fertilizer and the agronomic efficiency of P and K are not clear. Laub et al. (2023) analyze long-term trials of organic amendments that include manure, in combination with inorganic N sources. They find that over a long time horizon of 16 years, applications of manure or inorganic N alone result in declining yields, whereas the annual applications of a combination of manure and 120 kg/ha of inorganic fertilizer prevent yield decline, although there are some differences in the trend at different sites. This indicates the potential for organic fertilizers to prevent yield decline when used in combination with inorganic fertilizer.

Organic fertilizers contribute to improving soil health, but their nutrient release is slower compared to inorganic fertilizers. Combining organic and inorganic fertilizers generally results in higher yields and better nutrient content than using either type alone, as demonstrated by multiple studies. This integrated approach also helps maintain long-term soil fertility and crop productivity.



### 3. Heterogeneity in the impact of organic fertilizer on agronomic outcomes

#### Key takeaways:

- The effectiveness of organic fertilizers varies with soil fertility, type, management history, temperature, and precipitation.
- Different crops, varieties, and intercropping systems respond differently to organic fertilizers.
- The impact of organic fertilizers depends on the quantity and combination with inorganic fertilizers.
- The type and quality of organic fertilizers significantly influence their effectiveness on crop productivity and soil health.
- Higher quality organic fertilizers are more associated with short term yield improvements.

While substantial evidence demonstrates a positive impact of organic fertilizer (used individually or in combination with inorganic fertilizers) on nutrient quality, biomass and yields, the variability in outcomes highlights a need for context-specific recommendations for the usage of organic fertilizers, especially in application combined with inorganic fertilizers. This section discusses how the impact of organic fertilizers varies by agroecological conditions, crops and growing patterns, application rates, and organic fertilizer type and quality. Understanding these factors is crucial for creating localized recommendations that optimize the benefits of organic fertilizers.

#### 3.1 Agroecological conditions

Organic fertilizers improve soil quality by feeding nutrients to the soil, and their effectiveness depends on the baseline soil fertility, which is influenced by soil type and the plot's management history (Chivenge et al., 2011; Zingore et al., 2007). In addition, the efficiency of nutrient use, which comprises capture efficiency and conversion efficiency, varies with soil characteristics (Tittonell et al., 2008), and so applying the same amount of organic fertilizer to soils that have different nutrient levels could generate a variety of responses. Severely degraded soils may require multi-purpose solutions over multiple years to restore productivity, as seen in studies from Zimbabwe (Zingore et al., 2007, 2015). In a three-year study across plots of varying fertility, Zingore et al. (2007) found that application of inorganic fertilizers alone improves yields in the more fertile fields (by soil type and better past management) in the first year, but does not do so in less fertile fields. Application of organic fertilizer in the less fertile fields does not yield any impacts in the first two years, but results in significant increases in yields in the third year. They recommend targeted application of organic and inorganic fertilizers according to soil type and prior field management to improve crop yields and nutrient use efficiencies.



In addition to soil fertility, other agro-climatic factors, such as water availability and temperature, also play critical roles in the impact of organic fertilizers. For instance, a meta-analysis of 464 trials across 168 studies in China shows that temperature and precipitation affect the impact that organic fertilizers have on soil organic carbon content (Wang et al., 2024). They find that the impact of organic fertilizer on soil organic carbon is indistinguishable from that of inorganic fertilizer at high temperatures, but significantly higher in middle and low temperatures—this improvement effect of organic fertilizer is twice that of inorganic fertilizer in low-temperature areas, and 1.3 times in medium-temperature areas. Comparing organic fertilizer impact levels across temperatures also indicates that improvements in soil organic carbon are highest in the low-temperature regions, and lowest in the high-temperature regions. When it comes to precipitation, the authors find that the impact of organic fertilizer on soil organic carbon is 10.2 times higher than that of inorganic fertilizer in regions with between 400 mm rainfall annually, and 6.6 times higher in regions with more than 800 mm of rainfall annually.

Field studies also indicate that organic fertilizers, alone or combined with inorganic fertilizers, enhance yields under varying irrigation conditions. Examining the effect of crop production factors, namely water and N, Wang et al. (2020) run a three-year field study in China, and find that organic fertilizers alone lead to higher annual grain yields in the farming schemes without irrigation, while the combination of inorganic fertilizers and organic fertilizers lead to higher annual grain yields in the farming schemes without irrigation. To control the crop growing conditions even more closely, Anyega et al. (2021), a study mentioned above, investigate the performance of fertilizers under greenhouse and open-field conditions in Kenya. They find that adding organic fertilizer (i.e., frass) to inorganic fertilizer leads to significantly higher kale yields in the greenhouse setting. At the same time, they observe no statistical difference between different fertilizer applications in open-field conditions, highlighting the importance of controlled growing conditions.

#### 3.2 Crops and growing patterns

Different crops have distinct nutrient requirements and respond differently to fertilizers. For instance, a study applying fertilizers to French beans, tomatoes, and kale shows that yields improve significantly with organic fertilizer, though the degree of improvement varies by crop (Anyega et al., 2021). The researchers apply as much fertilizer on each experimental plot as would supply 100 percent of the N required, i.e., 371 kg/ha across crops and plots, and find that yields in fertilized soils are consistently higher for all crops compared with yields in unfertilized soils, but to different extents across crops. The application of one type of organic fertilizer (i.e., frass) improves the yield by 63–71 percent and 52–65 percent for tomato and French beans, respectively, compared with other fertilizer treatments. Cereal crops, like sorghum and maize, also respond differently to organic and inorganic fertilizers, with sorghum showing better results in some cases. In Amujoyegbe et al., (2007), the authors assess the effect of organic and inorganic fertilizer in a field trial in Nigeria and find that sorghum responds significantly better to the nutrient provision compared to maize. Wortmann et al. (2017) find some disparities in yield response to organic fertilizers in a global meta-analysis, with leafy crops showing the highest response and root/tuber crops showing the lowest response after a single application of organic fertilizer.

Even within the same crop, different varieties can have varied responses. Islam et al. (2017) conduct field trials in Bangladesh to examine the impact of fertilizers on two varieties of tomato. They find



that, compared to the control plots where no fertilizer is applied, yields in treatment plots where a combination of organic and inorganic fertilizer is applied increase slightly from 8.47 t/ha to 9.94 t/ha for one variety (i.e., BARI 15) and almost double from 10.45 th/ha to 20.84 t/ha for another variety (i.e., Roma VF).

Intercropping patterns also influence the performance of organic fertilizers. For example, maize yields in combination with common beans or soybeans improve significantly when both organic and inorganic fertilizers are used, though the magnitude of improvement varies by intercropping system (Chichongue et al., 2020). The authors record that, compared to the control plots, maize yields in plots treated with a combination of organic fertilizer (manure) and inorganic fertilizer (NPK) increase to 5.42 t/ha from 3.67 t/ha, and to 5.41 t/ha from 1.60 t/ha for maize-common bean and maize-soybean intercropping systems, respectively.

#### 3.3 Application rates

The quantity of fertilizer applied per unit of land is another factor contributing to the heterogeneity of organic fertilizer impacts. Optimal application rates depend on specific conditions, including soil type, crop, and yield targets; and so when farmers make different choices about application rates, whether to combine organic and inorganic fertilizers, etc., the impact of applied organic fertilizers on crop performance could be highly heterogeneous.

Studies with multiple fertilizer treatments generally find that a combination of organic and inorganic fertilizers yields the best results in terms of improved yields, or improved soil properties such as organic C and N content, soil sucrose, etc. (Kearney et al., 2012; Tovihoudji et al., 2018; Zhou et al., 2022). The treatment arms within these studies generally share the same level of nutrient inputs (i.e., N, P, K inputs), but these inputs are supplied by different fertilizer types and application amounts, such as pure organic fertilizer, pure inorganic fertilizer, 75 percent organic fertilizer and 25 percent inorganic fertilizer and 75 percent inorganic fertilizer.

Some experiments find that the specific nutrient sources (i.e., the division between organic fertilizer and inorganic fertilizer) do not significantly affect the crop outcomes, and could be determined by fertilizer prices and budget constraints (Kearney et al., 2012; Tovihoudji et al., 2018). On the other hand, some experiments find that nutrient input choices matter. Zhou et al. (2022) show that, compared with recommended inorganic fertilizer application, maize yields are higher only when inorganic fertilizer is coupled with a large amount of organic fertilizer (i.e., 45 t/ha). If the amount of organic fertilizer is a small (i.e., 15 t/ha) or medium (i.e., 30 t/ha) quantity, there is no significant difference in yields.

#### 3.4 Organic fertilizer type and quality

Organic fertilizers could be produced commercially or in the farmyard, and commonly come in the form of vermicompost, compost, and manure. The type and quality of organic fertilizers significantly influence their impact on crop productivity and soil health.

Different types of organic fertilizers are shown to affect these soil properties differently. Adekiya et al. (2020) observe that different types of manures applied to okra result in significantly different soil



parameters, including organic matter (OM), N, P, K, and magnesium, within one- and two-year horizons after application. More rapidly mineralizable manures, such as poultry manure, raise soil nutrient status more, whereas manures that are higher in lignin and those with higher C:N ratios raise OM more. Nevertheless, Mahmood et al. (2017) find that manures with higher C:N may stimulate a priming effect, which refers to added organic material leading to the loss of SOM previously existing in the soil. This is thought to be due to an increase in microbial populations caused by adding an organic substrate, which leads to a bigger population able to decompose the previously existing OM in the soil (Fontaine et al., 2004). However, even with the potential priming effect of higher C:N manures, SOM status is improved more by these treatments than by treatments without organic materials.

Looking at the impact of different organic fertilizers on downstream agricultural outcomes, Mahmood et al. (2017) compare the impact of three types of manure—farmyard manure, poultry manure (PM), and sheep manure—and conclude that PM is more effective at improving maize growth and yield than the other two types of manures. Masarirambi et al. (2010) evaluate the impact of chicken manure, cattle manure, and composted chicken manure on the growth of red lettuce. They, too, find that the type of fertilizer applied has a significant influence on crop growth. Chicken manure leads to better productivity, such as a higher number of leaves, higher leaf dry mass, larger plant height, and more yield, while composted chicken manure generates higher nutrient content in the plants, such as higher calcium, iron (Fe), and zinc (Zn) contents. A global meta-analysis by Wortman et al. (2017) similarly finds poultry manure having the highest average yield response, as compared to cattle or swine manure, or plant composts. Anyega et al. (2021) compare the impact of black soldier fly frass fertilizer (BSFFF), conventionally composted brewer's spent grain, and commercial organic fertilizer on the growth of vegetable crops, and discover that BSFFF has a better performance than the other two fertilizers. Islam et al. (2017) compare two types of organic fertilizers—vermicompost and compost—and do not find a significant difference in their impact on tomato growth.

Beyond the type of organic fertilizer, the quality of the fertilizer is a crucial determinant of the response of crops to the fertilizer application. Ogunlela et al., (2005) clearly state that the optimal amount of manure would vary depending on the quality of the manure. Only with organic fertilizers of intermediate quality (i.e., those consisting of manure, compost or organic materials with high N, high polyphenols or high lignins, as defined in Palm et al. (2001), do Vanlauwe et al. (2011) find an increase in maize yield and the agronomic efficiency of applied nutrients, on plots treated with a combination of organic and inorganic fertilizer relative to plots treated with inorganic fertilizer alone. Chivenge et al. (2011) find a general increase in yield with increased quality and increased N-content of organic fertilizer, when the organic fertilizer is used both alone and in combination with inorganic fertilizers.

Relying on linear mixed effect modeling in their meta-analysis, Gram et al. (2020) find that the significant positive effects on maize yields and on the agronomic efficiency of applied nutrients are more pronounced at higher N rates and for high-quality organic fertilizers, when combining organic and inorganic fertilizers. Their findings suggest that, when compared to low-quality organic fertilizer (low N, high lignins), high quality organic fertilizer not only increases yield and agronomic efficiency, but also results in a slower decline of agronomic efficiency with the N rate, a decreased inter-seasonal yield variability, and an increased SOC.



# 4. The impact of organic fertilizer on environmental outcomes

#### **KEY TAKEAWAYS:**

- Partial replacement of inorganic fertilizers with organic fertilizers reduces inorganic fertilizer application, which could decrease environmental pollution and health risks from nutrient runoff and leaching.
- Organic fertilizers could contribute to carbon sequestration, aiding climate change mitigation.
- Substituting imported inorganic fertilizer with locally-produced organic fertilizer could reduce GHG emissions through decreased international transportation and better management of local organic waste.

#### 4.1 Impacts due to reduction in inorganic fertilizer use

A combination of organic and inorganic fertilizers not only has the greatest agronomic impact, leading to improved soil as well as providing accessible nutrients during plant growth, but also allows farmers to reduce the quantity of inorganic fertilizers used without necessarily experiencing a reduction in yield. Inorganic fertilizers contribute to various forms of environmental degradation, and their use sometimes leads to pollution that is harmful to human health.

Nitrogen and phosphate leaching and runoff contribute to eutrophication, and nitrates that often leach into groundwater are a human health hazard, leading to methemoglobina-related diseases, and have potential links to various other health effects (Lubkowski, 2016). This is largely due to the solubility of inorganic fertilizers, which are designed to allow for rapid uptake by plants during periods of rapid growth that require high nutrient availability in soils. However, this high solubility results in rapid loss through leaching in the case of N, whereas fixation of P in soils leads to loss through erosion into waterways (Sileshi et al., 2019). Reducing the rate of release of soluble forms of N and P into soils is one approach to reducing these environmental losses (Lubkowski, 2016). Organic fertilizers contribute to reaching this goal of reducing environmental losses associated with fertilization.

Although nitrate leaching is not known as a major problem in sub-Saharan Africa at the moment, there are examples of higher groundwater nitrate levels in more heavily fertilized fields. Regardless of the current level of nitrate leaching in the region, improved nutrient recovery by crops provides both environmental benefits as well as benefits to farmers (Kihara et al., 2020). Some studies show that leaching is reduced when manures and inorganic fertilizers are applied together (Murwira, 1994; Nyamangara et al., 2003). This reduced leaching is thought to be due to the greater synchrony of nutrient-uptake by crops under management that uses combined organic and inorganic amendments (Kihara et al., 2022). However, in a review paper on the potential for organic farming systems to reduce N leaching, Kirchmann and Bergström (2001) find no significant difference in leaching per unit of N applied in organic farming systems when compared to conventional farming systems.



#### 4.2 Impacts due to organic fertilizer use

As previously discussed, one of the impacts of organic fertilizer on soil is improving its water retention capacity (Lynch & Bragg, 1985; Tisdall & Oades, 1982). This helps communities adapt to floods, by absorbing water and preventing the runoff of pollutants, and to droughts, by holding water for longer (Bondi et al., 2022; United States Environmental Protection Agency [EPA], 2024). In addition, increasing soil C content through the addition of organic fertilizers contributes to carbon sequestration in the soil, reducing CO<sub>2</sub> in the atmosphere, and contributing to climate change mitigation efforts (EPA, 2024; Villat & Nicholas, 2024; Wong et al., 2023).

However, if organic fertilizers, especially those derived from animal manure or waste, are not treated or composted properly, they may introduce pathogens into the soil. This could contaminate crops and water supplies, and pose a risk to human and animal health (Adegoke et al., 2016). These risks can be mitigated by careful composting, or the use of commercial organic fertilizers which employ advanced, standardized techniques (Ramos et al., 2021; Singh, 2012). Compared with less processed manures of similar feedstocks, certain commercially processed organic fertilizers reduce ammonia losses—for instance, Assefa & Tadesse (2019) find that pelleted manure has 97.8 percent lower ammonia emissions relative to unprocessed manure. This reduces environmental impacts such as eutrophication and acidification of aquatic environments that are caused by ammonia (Wyer et al., 2022), and represents a major N saving (Assefa & Tadesse, 2019).

Another aspect to consider in the environmental impacts of fertilizer use in agriculture is the resulting emission of greenhouse gasses (GHGs) such as methane (CH4), nitrous oxide (N2O), and carbon dioxide (CO<sub>2</sub>), which are major contributors to climate change (Shakoor et al., 2020; Shakoor et al., 2021). The use of organic fertilizers supports methane oxidation under aerobic conditions, thus lowering methane emissions, but stimulates methane production under anaerobic conditions, thus increasing methane emissions (Conrad, 2007). For instance, in a field trial in a wheat-corn cropping system in China, Gong et al. (2022) find that the use of organic fertilizers (microbial fertilizers) reduces N<sub>2</sub>O and CH<sub>4</sub> fluxes and reduces CO<sub>2</sub> emissions by 6.9-18.9 percent compared to inorganic fertilizers, by conserving soil heat and moisture during the corn season. In a field trial with maize in Zimbabwe across soil types and seasons, Mapanda et al. (2011) compare organic fertilizer (composted manure) with inorganic fertilizer and a control group, and find that organic fertilizer significantly reduces N<sub>2</sub>O emissions compared to the impact of inorganic fertilizers, especially during the wet season. However, the results for CH<sub>4</sub> and CO<sub>2</sub> fluxes were variable and dependent on soil water content. Similarly, in a global meta-analysis of 48 studies, Shakoor et al. (2021) show that the type of organic fertilizer, application rate, soil texture, climate zone, and crop type have an impact on the extent of GHG emissions. This suggests that the use of organic fertilizers does have the potential to reduce GHG emissions in the agricultural sector, although reducing GHG emissions from agricultural practices depends on management practices and use conditions.

#### 4.3 Impacts due to production of organic fertilizers

The improper disposal of organic wastes in landfills or unmanaged environments leads to anaerobic decomposition, releasing substantial amounts of methane, a GHG (EPA, 2024). Composting organic waste instead of landfilling can significantly lower methane emissions, as organic materials decompose aerobically rather than anaerobically, thus minimizing methane production (Brown et al., 2008;



Zhang et al., 2019; Li et al., 2023). Composting can also be responsive to local waste streams, further enhancing sustainability. Commercial production of organic fertilizer provides further opportunity for organic waste conversion into organic fertilizer for crop production, which could potentially reduce negative environmental impacts from existing waste management practices (Bidzakin et al., 2023).

The production and transportation of inorganic fertilizers are energy-intensive and release significant amounts of CO<sub>2</sub>, N<sub>2</sub>O, and CH<sub>4</sub> (Wood & Cowie, 2004; Smith et al., 2014). For instance, estimates indicate that the production and use of nitrogen fertilizers account for approximately five percent of global GHG emissions (Gao & Serrenho, 2023), while the synthesis of ammonia from which fertilizers are produced directly accounts for 10 percent of global GHG emissions and two percent of global energy (Gao & Serrenho, 2023). In Kenya, the use of inorganic fertilizers is responsible for one percent of Kenyan GHG emissions (0.6 Mt), and, in addition, the import of inorganic fertilizer causes further GHG emissions (3.4 Mt) upstream in countries where it is produced (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b). Substituting imported inorganic fertilizer with locally-produced organic fertilizer could reduce GHG emissions occurred during transportation. Local production of organic fertilizers could also potentially reduce GHG emissions through treating organic wastes properly, as discussed above.



# 5. The impact of organic fertilizer on economic outcomes

#### **KEY TAKEAWAYS:**

- Organic fertilizers lead to increase in income and profits, supported by evidence in the experimental settings from agronomic trials and in the real-life setting from survey data.
- There are differences in the impact levels of organic fertilizer observed in the experimental setting and in the real-life setting. Those discrepancies in productivity could go in either direction. Possible reasons include variations in soil characteristics and behavioral choices.

This section provides evidence of the economic benefits accrued from the application of organic fertilizers. The use of organic fertilizers, both alone and in combination with inorganic fertilizers, results in better soil health, increased yield, and higher nutritional quality of crops as shown in Section 2, and improved environmental benefits as shown in Section 4. Moving along the theory of change, this section investigates whether adoption of organic fertilizers could generate positive gains on smallholder farmers' livelihood, such as increasing income, profits, and food consumption, using both insights from agronomic trials and learnings in real-life settings.

#### 5.1. Economic impact in agronomic trials

Testing the potential of organic fertilizers, Kearney et al. (2012) conduct field trials on cabbage in Uganda. They find that, compared with control plots, the market value of cabbage heads, estimated by a group of local produce retailers, is increased in all treatment plots that use various fertilizer applications. The increase in economic benefit is 98 percent in the pure organic fertilizer (manure) treatment and 162-186 percent in the combination of organic and inorganic fertilizer treatment. Based on average land that farmers cultivate cabbage on in the region, the authors estimate that using a combination of organic and inorganic fertilizer alone. Exploring the benefits of partially substituting the chemical fertilizer with organic fertilizer, Zhai et al. (2022) conduct field trials on wheat in China, and observe increases in the net benefit by USD 336/ha and USD 208/ha for 15 percent and 30 percent substitution, respectively.

In terms of gross margin, Tovihoudji et al. (2018) examine the impact of organic fertilizer in combination with inorganic fertilizer on maize farming in Benin over a four-year period. The authors find that the gross margin increases 26-96 percent from applying manure, compared with not applying manure. Dada and Ewulo (2015) study maize farming in Nigeria, and find improved gross margins (i.e., USD 954) from combining poultry manure with superphosphate and urea, compared to no nutrient amendment (i.e., USD 245).



#### 5.2. Economic impact in a real-world setting

To reveal the impact in the real-world setting, another strand of studies conducts surveys with smallholder farmers, and estimates the yield differences between adopters and non-adopters of organic fertilizer, using quasi-experimental methods. Mwaura et al. (2021) interview 137 maize producers in Kenya and find that farmers who use either organic fertilizers (animal manure) solely or organic combined with inorganic fertilizers report higher gross margins, compared to farmers who use inorganic fertilizers solely. Those differences are statistically significant, and in the range of 16–80 percent higher for sole organic fertilizer users and 42–55 percent higher for combined inorganic and organic fertilizer users. Wainaina et al. (2018) conduct a representative survey with 1,344 maize farmers over all six agroecological zones in Kenya, to examine the impact of agricultural technology on household incomes. They observe that adopting organic fertilizer (manure) increases farm household income by 14 percent (USD 360) and per capita income by 20 percent (USD 99), compared to not adopting the organic fertilizer. By interviewing 300 farmers across three regions in Ghana, Bidzakin et al. (2023) find that organic fertilizer adoption increases yields by 57 percent, farm income by 53 percent, and farm gross margins by 63 percent.

Expanding the scope to the national level by comparing adopters and non-adopters in the sixth wave of the Ghana Living Standard Survey, Martey (2018) demonstrates that organic fertilizer (green and animal manure) increases crop productivity 1.43 times, increases crop income by USD 132, reduces total household expenditure by USD 174, reduces food expenditure by USD 58, and reduces poverty by 8 percent. In a similar study that uses data from the 2018/2019 Nigeria General Household Survey, Oyetunde-Usman et al. (2021) show that households that use organic fertilizer (green manure, compost, crop residue, and animal manure) increase their per capita asset value, per capita total house expenditure, and per capita food expenditure by USD 1,925, USD 407, and USD 236, respectively, compared to non-adopters. In the context of Ethiopia, Gelgo et al. (2017) collect survey data from 368 farmers in Shashemene district, and find that adoption of organic fertilizers increases farm income per hectare by USD 111–124, compared to non-adopters.

While those studies present significantly positive results, it is worth noting that the sample sizes used in some studies are small. More importantly, the economic analysis methods adopted, such as propensity score matching, rely on using statistical techniques to construct an artificial control group. More rigorous examinations on organic fertilizer in the real-world setting are needed to show its causal impacts.

### 5.3. Factors contributing to the differences in productivity between agronomic trials and real-world settings

When an agricultural input is adopted under real-life conditions, the benefit realized by smallholder farmers may differ from the gains observed in agronomic trials (Abate et al., 2018; Dar et al., 2013; Laajaj et al., 2020). The discrepancy in productivity could go in either direction. Possible reasons for the lower outcomes in the real-farm settings include differences in soil characteristics, and in farmers' behavioral choices and characteristics. A possible reason for higher outcomes in the real-farm settings investment.



Among the field trials that examine the impact of organic fertilizer, baseline soil properties are commonly measured, and the rate of organic fertilizer application is determined so as to supply the required nutrient content to that specific soil. On the other hand, in the real-life setting, smallholder farmers rarely conduct soil tests on their plots (Olwande, 2019; Wawire et al., 2021), and usually choose the application rate based on past experience or on general advisory suited for the aggregate location. These approximated rates may not be the optimal application rates of the farmers' particular cropland, and would thus decrease the potential impact of inputs. The issue is amplified when farmers choose to apply organic fertilizer based on the soil conditions, such as applying organic fertilizer to less fertile areas that would require additional nutrient supplement compared to the standard application rate.

Furthermore, smallholder farmers may face financial and logistical constraints when deciding about the application of a fertilizer. Bidzakin et al. (2023) show that the adoption of organic fertilizer is correlated with farmers' organization membership status and access to organic fertilizer, and with transportation costs in Ghana. Given those external constraints, farmers may adopt organic fertilizer at a suboptimal level even when they know the optimal rate.

Farmers' behavioral choices, such as in farming management and implementation, are also pivotal reasons for the disparity between experimental trials and real-world settings. One aspect centers on the correct adoption of organic fertilizer, such as when to apply and how to apply the organic fertilizer, in addition to the application rate discussed in section 3. Other aspects relate to other complementary agricultural practices, such as the type of seeds used, planting time, frequency of weeding, and usage of irrigation. Tittonell et al. (2008) show that there is a significant yield gap between farmer-and researcher-managed fields, due to differences in agronomic management and disparities in resource-use efficiencies. Because of the importance of appropriate management, supporting farmers with digital extension service, which could provide customized, timely, actionable advisory on both appropriate usage of organic fertilizer and other complementary practices, is an effective policy action that helps farmers reach their yield potential.

On the other hand, the application of organic fertilizers improves soil physicochemical and biological properties. The increased soil health results in higher efficiency and marginal returns of other input applications, such as improved seeds and inorganic fertilizer. As a consequence, farmers may crowd in additional investment on their farming, which in turn increases agricultural yield and income further (Boucher et al., 2021; Emerick et al., 2016).



## 6. Cost-effectiveness of organic fertilizer

#### Key takeaways:

- There is suggestive evidence indicating that the application of organic fertilizer is an economically viable choice, having a benefit-cost ratio that is bigger than 1. More rigorous insights are needed to provide further and complete affirmation, especially for commercial organic fertilizer.
- To maximize farmers' return of investment, it is also important to assess which organic fertilizer application offers the highest cost effectiveness.

The ultimate assessment of organic fertilizer is its cost-effectiveness for smallholder farmers. The inherent impact level of organic fertilizers on agricultural productivity is a key determinant of organic fertilizer's benefit-cost ratio. External market factors also play a crucial role. The benefit levels vary with the market value of the harvest. Given a fixed amount of harvest, if the market price of output is high, the total gross income farmers received would be large. In parallel, the cost levels vary with the input prices, which are affected by the availability and source of the organic fertilizer, be it commercially produced, or home-made compost, or manure.

Table 2 summarizes benefit-cost (b-c) ratios of the application of organic fertilizers or the combination of organic and inorganic fertilizers from seven studies. Benefit-cost ratios of organic fertilizers are defined as additional net income incurred from the organic fertilizer application divided by the additional cost of the organic fertilizer application. Benefit-cost ratios equaling one stand as a break-even scenario, the additional benefit being the same as the cost. If the b-c ratio is larger than one, it indicates that organic fertilizer is a cost-effective agricultural input. The few existing empirical studies generally demonstrate a benefit-cost ratio that is larger than one.

Focusing on non-commercially produced vermicompost and manure first, Kearney et al. (2012) perform a simple cost-benefit analysis on cabbage growing in Uganda. The authors calculate the benefit based on the market value of cabbage, and calculate the cost based on survey information for manure, and on the market price for inorganic fertilizer. They find that the combined use of organic and inorganic fertilizers results in net benefits that are 61 percent higher than use of inorganic fertilizer alone, and 39 percent higher than use of organic fertilizer alone. Tovihoudji et al. (2017) study the cost-effectiveness of maize farming in Benin, and how the benefit-cost ratios vary with the fluctuation of output and input prices. They observe that the benefit-cost ratios are always bigger than one (ranging between two and ten) in all experimental arms that have some amount of nutrient amendment, either manure or fertilizer or a combination of them. Even when there is a significant drop in the output price, i.e., decreasing the maize price by 25 percent, the application of organic fertilizer still remains financially attractive with b-c ratios bigger than two. Similarly, Jjagwe et al. (2020) estimate that the benefit-cost ratios for various manure usages on maize in Uganda are all bigger than one, ranging from 3 to 168, under the average interest rate (13.9 percent) assumption.



effectiveness. Commercial Agriculture for Smallholders and Agribusiness (CASA) report highlights the lack of scientific evidence on commercial products, and advocates for building a strong evidence base from large-scale trials (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b). Based on a few proprietary reports from a commercial organic fertilizer company, the benefit-cost ratios of one commercial organic fertilizer (i.e., Evergrow) are shown to be 3.35 for usage on French beans and bigger than 1 for various crops. In theory, the commercial processing of organic fertilizers can have greater cost effectiveness than less processed forms. Assefa and Tadesse (2019) find that the use of pelleted manures results in higher cost effectiveness per unit of nutrient applied, relative to similar unprocessed feedstocks, largely due to reduced N losses.

The evidence for the cost-effectiveness of organic fertilizer, especially commercial organic fertilizer, is sparse and calls for further impact evaluations to provide rigorous affirmation. While the limited available insights indicate the economic viability of organic fertilizer application in general, it is also essential for farmers to assess which organic fertilizer application offers the highest b-c ratio. Is it the home-produced organic fertilizer or commercially produced product, and is it when organic fertilizer is applied alone or when it is applied with inorganic fertilizer? Tovihoudji et al. (2017) find that the benefit-cost ratios of sole manure application rank on the top in three out of four seasons, compared to no fertilizer application and to the combination of manure and inorganic fertilizer. Mwaura et al. (2021) observe that sole animal manure application leads to the highest return in a sub-county in Kenya, while animal manure combined with inorganic fertilizer leads to the highest return in another sub-county. The answer to this optimum choice question would vary by context, depending on the local impact levels and costs.



## 7. Adoption of organic fertilizer

#### **KEY TAKEAWAYS:**

- Adoption of organic fertilizer remains low in Kenya and Africa.
- Several barriers must be addressed for the widespread adoption of organic fertilizer, such as providing information campaigns, extensive service, and financial assistance for farmers, and supporting the growth of the commercial organic fertilizer industry.

The usage of fertilizer has dramatically risen over the past several decades in Kenya, and Africa in general. Sub-Saharan Africa's fertilizer market has been growing at a rate of 8 percent per year since 2008 (Ariga et al., 2019). In Kenya, total fertilizer consumption doubled between 1990/1991 and 2011/2012 (World Bank, 2024) and reached 750,000 metric tons in 2021 (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b); this increase was mostly driven by inorganic fertilizer. The adoption rate of organic fertilizer remains low. In most regions of Kenya, less than 50 percent of farmers surveyed in the Tegemeo/MSU panel Household Survey report using organic fertilizers (Kibaara et al., 2008). Using the Living Standard Measurement Study–Integrated Surveys on Agriculture from six African countries (Ethiopia, Malawi, Niger, Nigeria, Tanzania, and Uganda), Sheahan and Barrett (2014) calculate that the proportion of cultivating households using organic fertilizer is only about 29 percent in 2010/2011. The usage of commercial organic fertilizer is even more sparse. According to the CASA report, only 1 percent of total fertilizer consumption is commercial organic products (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b).

Several barriers must be addressed for the widespread adoption of organic fertilizer; such barriers include insufficient knowledge, limited labor, financial constraints, and lack of policy support (Chianu et al., 2012). The first major cause of the low usage is inadequate knowledge of the application practices for and the impact of organic fertilizer. While the benefits of inorganic fertilizers-supplying nutrients to crops for improving their growth-have been assimilated, the gains of organic fertilizers are not well understood. Muluneh et al. (2022) and Okuma and Isiorhovoja (2017) highlight the effect of having information and experience about organic fertilizer on increasing its adoption significantly. Transitioning from locally-made manure and composting to commercially-produced organic fertilizer faces further knowledge challenges. In Kenya, manufactured organic fertilizer products are of varying quality, and strong evidence of their yield impact and standard recommendations for using them are not sufficiently generated (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b). Without widely recognized efficacy and usage guidelines, farmers are reluctant to adopt the product because of risk averseness and inadequate knowledge. Providing extension service and training agro-dealers could potentially close up those knowledge gaps and, in turn, drive up the adoption of commercially-produced organic fertilizer. In their examination of the organic fertilizer usage in Africa, various papers suggest increasing information campaigns and extension works to raise awareness about the importance of organic fertilizer, and thereby expanding the adoption of it among smallholder farmers (Fasina, 2013; Macharia et al., 2014; Muluneh et al., 2022; Mwaura et al., 2021; Okuma and Isiorhovoja, 2017).



In addition to the low demand from farmers, the supply of organic fertilizer has immediate challenges as well. One pivotal reason is the paucity of available organic resources (Bidzakin et al., 2023; Giller et al., 2009; Place et al., 2003). Not every farmer may be able to use their crop residues for composting, or have sufficient animal manure to apply to all their fields. When some raw materials exist, the amount of raw ingredients available are usually smaller than the amount needed to provide the required nutrients, and there are competing uses for these scarce resources, such as livestock feed or fuel (Anyega et al., 2021; Ndambi et al., 2019; Rufino et al., 2011). This barrier is especially salient for manure and home-made composting, which require a sizable application rate, typically thousands of kilograms per hectare. Gizaw and Kalsa (2017) show that increasing the availability of composting material could increase the adoption of organic fertilizer in Ethiopia. Commercial organic fertilizers offer a solution to the issues of insufficient organic resources and long composting time by using improved composting techniques and accessing large aggregated waste sources. But given the nascent commercial organic fertilizer industry in Kenya and sub-Saharan Africa, improvement of the production line (i.e., a more effective biomass supply chain and a higher local manufacturing capacity) and distribution channels is needed to improve the availability of product in all locations and at relevant agricultural timing.

Another vital advantage of commercial organic fertilizer compared to manure or home-made compost is its higher and more stable quality that has a more favorable C:N ratio and a lower risk of pathogens. Nonetheless, the higher quality comes with an inevitable hurdle-a higher cost. While the prices of commercial organic fertilizers are usually lower than the prices of inorganic fertilizers, greater volumes of organic fertilizer are needed when organic fertilizers are used alone. Thus, the price per unit of nutrient is higher with commercial organic fertilizer than with inorganic fertilizer. Also, the bulky nature and high moisture content of organic fertilizer increases its transportation cost. Odhiambo and Magandini (2008) and Aderinoye-Abdulwahab and Salami (2017) underscore that the high transportation costs is one of the prominent limitations to the use of organic fertilizer. Because of the financial constraints that smallholder farmers usually face, and the up-front high cost barriers, interventions such as input subsidies and transportation support could contribute to the spread of smallholder farmers using organic fertilizer. Furthermore, some studies demonstrate the significant yield benefit when adopting organic and inorganic fertilizers together, and small differences in the yield benefit for different combination rates, thereby suggesting that the application ratio between organic and inorganic fertilizers can be based on their prices (Kearney et al., 2012; Tovihoudji et al., 2018).



### 8. Macro perspective of commercial organic fertilizer sector

#### **KEY TAKEAWAYS:**

• The growth of the local commercial organic fertilizer sector has the potential to generate positive impacts on local employment and reduce the economic volatility caused by international shocks.

The Kenyan commercial organic fertilizer industry represents a USD 3 million value in 2022 and is predicted to grow to USD 45–75 million value in 2030 (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b). This enormous expansion is expected to generate a diverse set of benefits for the economy. This increased production and usage of commercial organic fertilizers could increase soil health and agricultural productivity, and reduce negative environmental impacts, as discussed in previous sections. Another direct impact from this sector growth is the increase in employment opportunities (Bidzakin et al., 2023). A substantial number of jobs would be created if those organic fertilizers are produced locally. The openings would happen not only in the fertilizer manufacturing sector, but also in the waste management sector and in distribution channels.

Furthermore, almost all inorganic fertilizers in Kenya are imported from a handful of international players. As a result, this heavy reliance on foreign trades makes Kenya's fertilizer market extremely sensitive to international events. For example, a large component of the fertilizer retail price is the Free on Board (FOB) price, and FOB increased by 100 percent after the outbreak of the war in Ukraine because the supply was suddenly reduced (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b). The Kenyan inorganic fertilizer market experienced both a skyrocketed fertilizer price and a hiatus in the supply chain. World Food Programme & Boston Consulting Group (2022) estimates that fertilizer prices have risen by approximately 70 percent from 2021 to 2022. Substituting some inorganic fertilizers with commercial organic fertilizers applications, and producing those commercial organic fertilizers within the country, would alleviate the Kenyan fertilizer market's vulnerability to foreign shocks, such as international trade disruption and exchange rate fluctuations.

These macro perspectives are mainly drawn from one study conducted by Commercial Agriculture for Smallholders and Agribusiness using 60 in-country interviews and desk research and analysis to assess the fertilizer sector in Kenya (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b). Additional empirical evidence is called for understanding the full impact of sector growth.



## 9. Conclusion

Even though the agriculture sector is one of the main contributors in the economies of sub-Saharan Africa, the sector still faces a significant yield gap. Declining soil fertility and low productivity in current cropping systems are two key challenges. This study focuses on examining one potential solution that is often neglected—the use of organic fertilizer.

Applying organic fertilizer can result in increased SOM and an improved nutrient status of soil, and contributes to reducing soil degradation. There is strong evidence that, by improving the physical and chemical features of soil, organic fertilizer also improves the efficiency of other nutrient applications. Existing studies provide rigorous evidence that organic fertilizer application leads to significant benefits. Applying organic fertilizer increases agricultural productivity, profitability, and the nutrient values of the harvest. Additionally, organic fertilizer application, such as leaching, gaseous losses, and runoff. Moreover, the improved productivity and efficiency in resource use could crowd in additional agricultural investments, which increase yield and profit further. It is also important to note that the impact of organic fertilizer could vary substantially across different circumstances. Specifically, agroecological conditions, target crops and growing patterns, fertilizer application rates, and the type and quality of the organic fertilizer all influence the performance of organic fertilizers.

Despite the well-documented agronomic gains of organic fertilizer application, its adoption level remains low in Africa. Crucial barriers are the scarcity of raw organic material, local manufacturing capacity, and insufficient awareness of the benefits of organic fertilizer. Commercial organic fertilizer presents an opportunity to improve the usage of organic fertilizers significantly. Commercial production addresses the issues of insufficient organic resources and long composting time that home-made production commonly experiences. Using commercial organic fertilizer also has the potential to reduce negative environmental impacts through avoiding the environmental issues associated with utilizing inorganic fertilizer and organic waste.

Nevertheless, the additional costs associated with using commercial organic fertilizer, compared to using home-made compost or manure, are new challenges to be solved. Encouraging investment into commercial production of organic fertilizer could increase the supply of organic fertilizer and reduce the costs, thereby increasing farmers' access. Expanding the market provision could also reduce transportation costs, as products become available even in remote areas. Finally, coupling the sale of organic fertilizer with digital extension services could provide farmers with knowledge of the appropriate application, and therefore increase the yield and profit gain for farmers, making the adoption more cost effective. For an innovation to spread widely, agronomic gains are not sufficient—it has to be cost-effective for farmers. While there is suggestive evidence that indicates organic fertilizers are cost-effective, future impact evaluations that provide rigorous proof of the cost-effectiveness of commercial organic fertilizer are needed.



## 10. Figure and Tables

Figure 1: Framework for impact pathways of organic fertilizer



#### Table 1: Summary of various fertilizers

|                                    | Inorganic Fertilizer  | Organic Fertilizer  |
|------------------------------------|---|---|
| Main Purpose                       | Provide essential nutrients to crops;<br>improve crop yield immediately                   | Provide essential nutrients to crops;<br>provide carbon-based material to soils;<br>improve long-term soil fertility      |
| Application Rates                  | Typically tens to hundreds of kilograms per hectare                                       | Typically thousands of kilograms per hectare  |
| Common Sources                     | Synthesized from air; processed from mined minerals                                       | Animal manures; composted plant residues; animal byproducts   |
| Environmental / Health<br>Concerns | Leaching into groundwater; airborne particulates; eutrophication                          | Potential source of pathogens   |
| Challenges with<br>Adoption        | High cost; availability; need for<br>technical knowledge to select ideal type<br>and rate | Insufficient knowledge; lack of suffi-<br>cient quantities; tradeoffs in use of<br>plant materials; financial constraints |



#### Table 2: Cost-effectiveness of organic fertilizer

| Study  | Study Location     | Study<br>Crop                 | Study Year | Treatment  | Benefit-<br>Cost Ratio |      |  |
|--|--------------------|-------------------------------|------------|--|------------------------|------|--|
| Panel A: Manure and compost (farm trial data)                        |                    |                               |            |  |                        |      |  |
| Jjagwe et al.<br>(2020)  | Uganda             | maize                         | 2018       | manure with various storage methods <sup>1</sup>                   | 3.00-168.00            |      |  |
|  |                    |                               |            | vermicompost   | 102.00                 |      |  |
| Kearney et al.<br>(2012)   | Uganda             | cabbage                       | 2010       | manure   | 4.59                   |      |  |
| Tovihoudji et al.<br>(2017)  | Benin              | maize                         | 2012-2015  | manure with various application rates <sup>2</sup>                 | 2.00-10.00             |      |  |
| Panel B: Manure (collected survey data)                              |                    |                               |            |  |                        |      |  |
| Mwaura et al.<br>(2021)  | Kenya - Gatanga    | maize                         | maize 2018 | 2018   | sole animal manure     | 2.69 |  |
|  | Kenya - Meru South |                               |            | sole animal manure   | 1.63                   |      |  |
| Panel C: Commercial organic fertilizer (proprietary farm trial data) |                    |                               |            |  |                        |      |  |
| FarmStar report  | Kenya              | French<br>beans               | 2014       | commercial organic fertilizer<br>(Evergrow) + inorganic fertilizer | 3.35                   |      |  |
| SGS report   | Kenya              | tomato                        | 2024       | commercial organic fertilizer<br>(Evergrow) + inorganic fertilizer | >1                     |      |  |
| CropNuts report  | Kenya              | various<br>crops <sup>3</sup> | 2023       | commercial organic fertilizer<br>(Evergrow) + inorganic fertilizer | >1                     |      |  |

#### Note:

1. Various storage methods refer to storing under a shade, storing in the open, and storing in a biogas digester.

2. Various application rates refer to 3 Mg/ha and 6 Mg/ha.

3. Various crops include maize, rice, potato, onion, cabbage, and black nightshade.

## Reference



60 Decibels. (2023). Regen Organics: Impact performance report. 60 Decibels.

- Abate, G. T., Bernard, T., de Brauw, A., & Minot, N. (2018). The impact of the use of new technologies on farmers' wheat yield in Ethiopia: Evidence from a randomized control trial. *Agricultural Economics*, *49*(4), 409–421. https://doi.org/10.1111/agec.12425
- Adegoke, A., Awolusi, O., & Stenström, T. (2016). Organic fertilizers: Public health intricacies. In M. L. Larramendy & S. Soloneski, S. (Eds), *Organic fertilizers From basic concepts to applied outcomes*. InTech. https://doi.org/10.5772/64195
- Adekiya, A. O., Ejue, W. S., Olayanju, A., Dunsin, O., Aboyeji, C. M., Aremu, C., Adegbite, K., & Akinpelu, O. (2020). Different organic manure sources and NPK fertilizer on soil chemical properties, growth, yield and quality of okra. *Scientific Reports*, *10*(1), 16083. https://doi.org/10.1038/s41598-020-73291-x
- Aderinoye-Abdulwahab, S. A., & Salami, S. T. (2017). Assessment of organic fertilizer usage by vegetable farmers in Asa Local Government area of Kwara State, Nigeria. *Agrosearch*, *17*(1). https://doi. org/10.4314/agrosh.v17i1.8
- African Union. (2006). Abuja Declaration on fertilizer for the African green revolution—Africa Fertilizer Summit. African Union.
- African Union. (2024). *Draft Nairobi Declaration on Africa Fertilizer and Soil Health Summit*. African Union. https://au.int/sites/default/files/documents/43771-doc-AFSHS\_Decl\_4\_II\_Rev\_2\_E.pdf
- Ahmad, A. A., Radovich, T. J. K., Nguyen, H. V., Uyeda, J., Arakaki, A., Cadby, J., Paull, R., Sugano, J., & Teves, G. (2016). Use of organic fertilizers to enhance soil fertility, plant growth, and yield in a tropical environment. In M. L. Larramendy & S. Soloneski (Eds), Organic Fertilizers—From Basic Concepts to Applied Outcomes. IntechOpen. https://doi.org/10.5772/62529
- Amujoyegbe, B. J., Opabode, J. T., & Olayinka, A. (2007). Effect of organic and inorganic fertilizer on yield and chlorophyll content of maize (Zea mays L.) and sorghum *Sorghum bicolour* (L.) Moench). *African Journal of Biotechnology*, 6(16). https://doi.org/10.5897/AJB2007.000-2278
- Anyega, A. O., Korir, N. K., Beesigamukama, D., Changeh, G. J., Nkoba, K., Subramanian, S., Van Loon, J. J. A., Dicke, M., & Tanga, C. M. (2021). Black soldier fly-composted organic fertilizer enhances growth, yield, and nutrient quality of three key vegetable crops in Sub-Saharan Africa. *Frontiers in Plant Science*, 12, 680312. https://doi.org/10.3389/fpls.2021.680312
- Ariga, J., & Jayne, T. S. (2011). Fertilizer in Kenya: Factors driving the increase in usage by smallholder farmers. In P. Chuhan-Pole & M. Angwafo (Eds), Yes, Africa can: Success stories from a dynamic continent. The World Bank. https://elibrary.worldbank.org/doi/pdf/10.1596/978-0-8213-8745-0
- Ariga, J., Mabaya, E., Waithaka, M., & Wanzala-Mlobela, M. (2019). Can improved agricultural technologies spur a green revolution in Africa? A multicountry analysis of seed and fertilizer delivery systems. *Agricultural Economics*, 50(S1), 63–74. https://doi.org/10.1111/agec.12533
- Assefa, S., & Tadesse, S. (2019). The principal role of organic fertilizer on soil properties and agricultural productivity—A review. *Agricultural Research & Technology: Open Access Journal, 22*(2), 1–5. https://doi.org/10.19080/ART0AJ.2019.22.556192
- Beesigamukama, D., Mochoge, B., Korir, N., Fiaboe, K. K. M., Nakimbugwe, D., Khamis, F. M., Subramanian, S., Dubois, T., Musyoka, M. W., Ekesi, S., Kelemu, S., & Tanga, C. M. (2020). Exploring black soldier fly frass as novel fertilizer for improved growth yield, and nitrogen use efficiency of maize under field conditions. *Frontiers in Plant Science*, *11*.
- Beesigamukama, D., Mochoge, B., Korir, N., Musyoka, M. W., Fiaboe, K. K. M., Nakimbugwe, D., Khamis, F. M., Subramanian, S., Dubois, T., Ekesi, S., & Tanga, C. M. (2020). Nitrogen fertilizer equivalence of black soldier fly frass fertilizer and synchrony of nitrogen mineralization for maize production. *Agronomy*, 10(9), 1395. https://doi.org/10.3390/agronomy10091395
- Bekunda, M., Ebanyat, P., Nkonya, E., Mugendi, D., & Msaky, J. (2005). Soil fertility status, management,

and research in East Africa. *Eastern Africa Journal of Rural Development*, 20(1), 94–112. https://doi.org/10.4314/eajrd.v20i1.28362

- Bhanwaria, R., Singh, B., & Musarella, C. M. (2022). Effect of organic manure and moisture regimes on soil physiochemical properties, microbial biomass C<sub>mic</sub>:N<sub>mic</sub>:P<sub>mic</sub> turnover and yield of mustard grains in arid climate. *Plants*, 11(6), 722. https://doi.org/10.3390/plants11060722
- Bidzakin, J. K., Graves, A., Awunyo-Vitor, D., Yeboah, O., Yahaya, I., & Wahaga, E. (2023). Utilization of organic fertilizer in Ghana: Implications for crop performance and commercialization. *Advances in Agriculture*, 2023(4), 1–16. https://doi.org/10.1155/2023/8540278
- Birch, I. (2018). *Agricultural productivity in Kenya: Barriers and opportunities*. K4D Helpdesk Report. Institute of Development Studies. https://www.gov.uk/research-for-development-outputs/agricultural-productivity-in-kenya-barriers-and-opportunities
- Bondi, C., Castellini, M., & Lovino, M. (2022). Compost amendment impact on soil physical quality estimated from hysteretic water retention curve. *Water, 14*(7). https://www.mdpi.com/2073-4441/14/7/1002
- Boston Consulting Group. (2023). Farmer & agrovet survey compendium Key analysis & insights from survey results [Slides]. Boston Consulting Group.
- Boucher, S. R., Carter, M. R., Flatnes, J. E., Lybbert, T. J., Malacarne, J. G., Marenya, P., & Paul, L. A. (2021). Bundling genetic and financial technologies for more resilient and productive small-scale agriculture. *NBER Working Paper* No. 29234. National Bureau of Economic Research. https://doi. org/10.3386/w29234
- Brown, S., Kruger, C., & Subler, S. (2008). Greenhouse gas balance for composting operations. *Journal of Environmental Quality*, *37*(4):1396-1410.
- Chenu, C., Rumpel, C., Védère, C., & Barré, P. (2024). Methods for studying soil organic matter: Nature, dynamics, spatial accessibility, and interactions with minerals. In Soil microbiology, ecology and biochemistry (pp. 369–406). Elsevier.
- Chianu, J. N., Chianu, J. N., & Mairura, F. (2012). Organic fertilizers in sub-Saharan farming systems. In E. Lichtfouse (Ed.), *Organic fertilisation, soil quality and human health* (pp. 31–65). Springer Netherlands. https://doi.org/10.1007/978-94-007-4113-3\_3
- Chichongue, O., Karuku, G., & Magalhaes, A. (2020). Effect of organic and low inorganic fertilizers on yields in a maize-legume intercropping systems in two selected sites in Mozambique. *International Journal of Ecology and Ecosolution*, 7(2), 8–15. https://doi.org/10.30918/IJEE.72.20.013
- Chivenge, P., Sharma, S., Bunquin, M. A., & Hellin, J. (2021). Improving nitrogen use efficiency—A key for sustainable rice production systems. *Frontiers in Sustainable Food Systems*, *5*, 737412. https://doi.org/10.3389/fsufs.2021.737412
- Chivenge, P., Vanlauwe, B., & Six, J. (2011). Does the combined application of organic and mineral nutrient sources influence maize productivity? A meta-analysis. *Plant and Soil*, 342(1–2), 1–30. https://doi.org/10.1007/s11104-010-0626-5
- Commercial Agriculture for Smallholders and Agribusiness. (2023a). *Exploring fertilizer alternatives and opportunities for improving use efficiency in Africa and South Asia*. Commercial Agriculture for Smallholders and Agribusiness. https://www.casaprogramme.com/wp-content/uploads/2023/10/exploring-fertilizer-alternatives-and-opportunities.pdf
- Commercial Agriculture for Smallholders and Agribusiness. (2023b). *Scalable alternatives to inorganic fertiliser in Kenya*. Commercial Agriculture for Smallholders and Agribusiness. https://www.technoserve.org/wp-content/uploads/2023/11/FCDO\_Kenya\_Fertilizer\_Final\_Report\_vFF.pdf
- Conrad, R. (2007). Microbial ecology of methanogens and methanotrophs. *Advances in Agronomy*, 96, 1–63.
- Dada, B. F., & Ewulo, B. S. (2015). Economic returns of integrated use of poultry manure and inorganic fertilizers on maize yield in Akure, Nigeria. *International Journal of Agriculture Innovations and Research*, *3*(5).

- D'Alessandro, S. P., Caballero, J., Lichte, J., & Simpkin, S. (2015). Agricultural sector risk assessment, *World Bank Group Report No. 97887*. Agriculture Global Practice Technical Assistance Paper. World Bank Group.
- Dar, M. H., de Janvry, A., Emerick, K., Raitzer, D., & Sadoulet, E. (2013). Flood-tolerant rice reduces yield variability and raises expected yield, differentially benefitting socially disadvantaged groups. *Scientific Reports*, *3*(1), 3315. https://doi.org/10.1038/srep03315
- Dimkpa, C., Adzawla, W., Pandey, R., Atakora, W. K., Kouame, A. K., Jemo, M., & Bindraban, P. S. (2023). Fertilizers for food and nutrition security in sub-Saharan Africa: An overview of soil health implications. *Frontiers in Soil Science*, *3*. https://doi.org/10.3389/fsoil.2023.1123931
- Emerick, K., de Janvry, A., Sadoulet, E., & Dar, M. H. (2016). Technological innovations, downside risk, and the modernization of agriculture. *American Economic Review*, *106*(6), 1537–1561. https://doi.org/10.1257/aer.20150474
- Fan, H., Zhang, Y., Li, J., Jiang, J., Waheed, A., Wang, S., Rasheed, S. M., Zhang, L., & Zhang, R. (2023). Effects of organic fertilizer supply on soil properties, tomato yield, and fruit quality: A global meta-analysis. Sustainability, 15(3), 2556. https://doi.org/10.3390/su15032556
- Fasina, O. O. (2013). Determinants of perceived effectiveness of organic fertilizer used by farmers in Oyo State, Nigeria. *Agricultura Tropica et Subtropica*, *46*(1), 23–28. https://doi.org/10.2478/ats-2013-0004
- Fontaine, S., Bardoux, G., Abbadie, L., & Mariotti, A. (2004). Carbon input to soil may decrease soil carbon content. *Ecology Letters*, 7(4), 314–320. https://doi.org/10.1111/j.1461-0248.2004.00579.x
- Fonte, S. J., Yeboah, E., Ofori, P., Quansah, G.W., Vanlauwe, B., & Six, J (2009). Fertilizer and residue quality effects on organic matter stabilization in soil aggregates. *Soil Science Society of America Journal*, 73(3): 961-966.
- Food and Agriculture Organization [FAO]. (2018). Update on the global soil partnership, including the international code of conduct for the use and management of fertilizers. FAO Committee on Agriculture.
- Food and Agriculture Organization [FAO]. (2022). *Soils for nutrition: State of the art*. Food and Agriculture Organization. https://doi.org/10.4060/cc0900en
- Food and Agriculture Organization [FAO]. (2023). *Strengthening digital agricultural extension and advisory services in smallholder farming*. Food and Agriculture Organization. https://doi.org/10.4060/cc6267en
- Food and Agriculture Organization. (2024). *Kenya at a glance*. Food and Agriculture Organization. https:// www.fao.org/kenya/fao-in-kenya/kenya-at-a-glance/en/
- Forge, T., Hogue, E., Neilsen, G., & Neilsen, D. (2003). Effects of organic mulches on soil microfauna in the root zone of apple: Implications for nutrient fluxes and functional diversity of the soil food web. *Applied Soil Ecology*, *22*(1), 39–54.
- Franzluebbers, A. J. (2002). Water infiltration and soil structure related to organic matter and its stratification with depth. *Soil and Tillage Research*, 66(2), 197–205. https://doi.org/10.1016/S0167-1987(02)00027-2.
- Gao, Y., & Serrenho, A.C. (2023). Greenhouse gas emissions from nitrogen fertilizers could be reduced by up to one-fifth of current levels by 2050 with combined interventions. *Nature Food*, 4(2), 170–178.
- doi: 10.1038/s43016-023-00698-w. Epub 2023 Feb 9.
- Gelgo, B., Mshenga, P. M., & Zemedu, L. (2017). Analysis of the impact of organic fertilizer use on smallholder farmers' income in Shashemene district, Ethiopia. *International Journal of Agricultural Economics*, 1(4).
- Gicheru, P. (2012). An overview of soil fertility management, maintenance, and productivity in Kenya. *Archives of Agronomy and Soil Science*, *58*(1), S22–S32.
- Giller, K. E. (2020). The food security conundrum of sub-Saharan Africa. *Global Food Security*, 26, 100431. https://doi.org/10.1016/j.gfs.2020.100431
- Giller, K. E., Witter, E., Corbeels, M., & Tittonell, P. (2009). Conservation agriculture and smallholder farming in Africa: The heretics' view. *Field Crops Research*, *114*(1), 23–34. https://doi.org/10.1016/j. fcr.2009.06.017



- Gizaw, A. A., & Kalsa, K. K. (2017). Building a resilient wheat seed system to resist rust risks in Ethiopia. Journal of Life Sciences, 11(1). https://doi.org/10.17265/1934-7391/2017.01.003
- Global Yield Gap Atlas. (2024). *Global Yield Gap and Water Productivity Atlas*. www.yieldgap.org, May 1, 2024.
- Gong, H., Li, J., Liu, Z., Zhang, Y., Hou, R., & Ouyang, Z. (2022). Mitigated greenhouse gas emissions in cropping systems by organic fertilizer and tillage management. *Land*, *11*(7), 1026.
- Gram, G., Roobroeck, D., Pypers, P., Six, J., Merckx, R., & Vanlauwe, B. (2020). Combining organic and mineral fertilizers as a climate-smart integrated soil fertility management practice in sub-Saharan Africa: A meta-analysis. *PLOS ONE*, *15*(9), e0239552. https://doi.org/10.1371/journal.pone.0239552
- Grassini, P., Cassman, K. G., & Ittersum, M. K. van. (2017). Exploring maize intensification with the global yield gap atlas. *Better Crops with Plant Food*, 101(2), 7–9.
- Houben, D., Daoulas, G., Faucon, M.-P., & Dulaurent, A.-M. (2020). Potential use of mealworm frass as a fertilizer: Impact on crop growth and soil properties. *Scientific Reports*, *10*(1), 4659. https://doi.org/10.1038/s41598-020-61765-x
- International Organization for Standardization. (2022). *Fertilizers, soil conditioners and beneficial substances–Vocabulary* (ISO 8157:2022). International Organization for Standardization. https://www.iso.org/standard/80949.html
- Islam, M., Islam, S., Akter, A., Rahman, M., & Nandwani, D. (2017). Effect of organic and inorganic fertilizers on soil properties and the growth, yield and quality of tomato in Mymensingh, Bangladesh. *Agriculture*, 7(3), 18. https://doi.org/10.3390/agriculture7030018
- Jayne, T. S., Mason, N. M., Burke, W. J., & Ariga, J. (2018). Review: Taking stock of Africa's second-generation agricultural input subsidy programs. *Food Policy*, *75*, 1–14. https://doi. org/10.1016/j.foodpol.2018.01.003
- Jjagwe, J., Chelimo, K., Karungi, J., Komakech, A. J., & Lederer, J. (2020). Comparative performance of organic fertilizers in maize (*Zea mays* L.) growth, yield, and economic results. *Agronomy*, *10*(1), 69. https://doi.org/10.3390/agronomy10010069
- Karlyn, A. (2017). Comparative analysis of smallholder farmers in Kenya, Zambia, and Tanzania. MercyCorps and Nathan.
- Kay, B., & Angers, D. (2001). Soil structure. In A. W. Warrick. (Ed.), *Soil physics companion* pp. 249–295. CRC Press.
- Kay, B., da Silva, A. P., & Baddock, J. A. (1997). Sensitivity of soil structure to changes in organic carbon content: Predictions using pedotransfer functions. *Canadian Journal of Soil Science*, 77, 655–667.
- Kearney. (2023). Sanergy (Regen Organics) social performance evaluation using SROI methodology [Slides]. Kearney.
- Kearney, S., Fonte, S. J., Salomon, A., Six, J., & Scow, K. M. (2012). Forty percent revenue increase by combining organic and mineral nutrient amendments in Ugandan smallholder market vegetable production. *Agronomy for Sustainable Development*, 32(4), 831–839. https://doi.org/10.1007/ s13593-012-0097-6
- Kenya Bureau of Standards [KEBS]. (2023). KS 2290: 2023, Organic fertilizer—Specification. Kenya Bureau of Standards.
- Kenya National Bureau of Statistics [KNBS]. (2023). *Economic survey 2023*. Kenya National Bureau of Statistics. https://www.knbs.or.ke/download/economic-survey-2023/
- Kibaara, B., Ariga, J., Olwande, J., & Jayne, T. S.(2008). Trends in Kenyan agricultural productivity: 1997-2007. *Tegemeo Institute of Agricultural Policy and Development Working Paper 31/2008*. Tegemeo Institute, Egerton University. https://doi.org/10.22004/ag.econ.56117
- Kihara, J., Bolo, P., Kinyua, M., Nyawira, S. S., & Sommer, R. (2020). Soil health and ecosystem services: Lessons from sub-Sahara Africa (SSA). *Geoderma*, 370, 114342. https://doi.org/10.1016/j. geoderma.2020.114342

- Kihara, J., Manda, J., Kimaro, A., Swai, E., Mutungi, C., Kinyua, M., Okori, P., Fischer, G., Kizito, F.,
  & Bekunda, M. (2022). Contributions of integrated soil fertility management (ISFM) to various sustainable intensification impact domains in Tanzania. *Agricultural Systems*, 203, 103496. https://doi.org/10.1016/j.agsy.2022.103496
- Kihara, J., Nziguheba, G., Zingore, S., Coulibaly, A., Esilaba, A., Kabambe, V., Njoroge, S., Palm, C., & Huising, J. (2016). Understanding variability in crop response to fertilizer and amendments in sub-Saharan Africa. *Agriculture, Ecosystems & Environment, 229*, 1–12. https://doi.org/10.1016/j. agee.2016.05.012
- Kimetu, J. M., Lehmann, J., Ngoze, S. O., Mugendi, D. N., Kinyangi, J. M., Riha, S., Verchot, L., Recha, J. W., & Pell, A. N. (2008). Reversibility of soil productivity decline with organic matter of differing quality along a degradation gradient. *Ecosystems*, *11*(5), 726–739. https://doi.org/10.1007/s10021-008-9154-z
- Kirchmann, H., & Bergström, L. (2001). Do organic farming practices reduce nitrate leaching? *Communications in Soil Science and Plant Analysis*, 32(7–8), 997–1028. https://doi.org/10.1081/ CSS-100104101
- Kogo, B. K., Kumar, L., & Koech, R. (2020). Impact of land use/cover changes on soil erosion in western Kenya. *Sustainability*, *12*(22), 9740. https://doi.org/10.3390/su12229740
- Kopittke, P. M., Dalal, R. C., Finn, D., & Menzies, N. W. (2017). Global changes in soil stocks of carbon, nitrogen, phosphorus, and sulphur as influenced by long-term agricultural production. *Global Change Biology*, *23*(6), 2509–2519. https://doi.org/10.1111/gcb.13513
- Kopittke, P. M., Dalal, R. C., Hoeschen, C., Li, C., Menzies, N. W., & Mueller, C. W. (2020). Soil organic matter is stabilized by organo-mineral associations through two key processes: The role of the carbon to nitrogen ratio. *Geoderma*, *357*, 113974. https://doi.org/10.1016/j.geoderma.2019.113974
- Laajaj, R., Macours, K., Masso, C., Thuita, M., & Vanlauwe, B. (2020). Reconciling yield gains in agronomic trials with returns under African smallholder conditions. *Scientific Reports, 10*(1), 14286. https://doi.org/10.1038/s41598-020-71155-y
- Lal, R. (2006). Enhancing crop yields in the developing countries through restoration of the soil organic carbon pool in agricultural lands. *Land Degradation & Development*, *17*(2), 197–209. https://doi. org/10.1002/ldr.696
- Lassaletta, L., Billen, G., Grizzetti, B., Anglade, J., & Garnier, J. (2014). 50 year trends in nitrogen use efficiency of world cropping systems: The relationship between yield and nitrogen input to cropland. *Environmental Research Letters*, 9(10). https://iopscience.iop.org/article/10.1088/1748-9326/9/10/105011/meta
- Laub, M., Corbeels, M., Mathu Ndungu, S., Mucheru-Muna, M. W., Mugendi, D., Necpalova, M., Van de Broek, M., Waswa, W., Vanlauwe, B., & Six, J. (2023). Combining manure with mineral N fertilizer maintains maize yields: Evidence from four long-term experiments in Kenya. *Field Crops Research*, 291, 108788. https://doi.org/10.1016/j.fcr.2022.108788
- Lazcano, C., Zhu-Barker, X., & Decock, C. (2021). Effects of organic fertilizers on the soil microorganisms responsible for N<sub>2</sub>O emissions: A review. *Microorganisms*, 9(5), 983. https://doi.org/10.3390/microorganisms9050983
- Leroy, B. (2008). Soil food web, C and N transformations and soil structure: Interactions and feedback mechanisms as a function of the quality of exogenous organic matter. PhD Thesis, Ghent University. http://hdl.handle.net/1854/LU-468667
- Leroy, B., Herath, H., Sleutel, S., De Neve, S., Gabriels, D., Reheul, D., & Moens, M. (2008). The quality of exogenous organic matter: Short-term effects on soil physical properties and soil organic matter fractions. *Soil Use and Management*, 24(2), 139–147. https://doi.org/10.1111/j.1475-2743.2008.00142.x
- Li, M., Li, S., Chen, S., Meng, Q., Wang, Y., Yang, W., Shi, L., Ding, F., Zhu, J., Ma, R., & Guo, X. (2023). Measures for controlling gaseous emissions during composting: A Review. *International Journal of Environmental Research and Public Health*, 20(4), 3587.

- Li, S., Li, J., Zhang, B., Li, D., Li, G., & Li, Y. (2017). Effect of different organic fertilizers application on growth and environmental risk of nitrate under a vegetable field. *Scientific Reports*, 7(1), 17020. https://doi.org/10.1038/s41598-017-17219-y
- Lin, S., Lei, Q., Liu, Y., Zhao, Y., Su, L., Wang, Q., Tao, W., & Deng, M. (2023). Quantifying the impact of organic fertilizers on soil quality under varied irrigation water sources. *Water*, *15*(20), 3618. https://doi.org/10.3390/w15203618
- Lori, M., Symnaczik, S., Mäder, P., Deyn, G. D., & Gattinger, A. (2017). Organic farming enhances soil microbial abundance and activity—A meta-analysis and meta-regression. *PLOS ONE*, *12*(7), e0180442. https://doi.org/10.1371/journal.pone.0180442
- Lubkowski, K. (2016). Environmental impact of fertilizer use and slow release of mineral nutrients as a response to this challenge. *Polish Journal of Chemical Technology*, *18*(1), 72–79. https://doi. org/10.1515/pjct-2016-0012
- Lynch, J. M., & Bragg, E. (1985). Microorganisms and soil aggregate stability. In B. A. Stewart (Ed.), *Advances in soil science* (pp. 133–171). Springer. https://doi.org/10.1007/978-1-4612-5088-3\_3
- Macharia, J., Mugwe, J., Mucheru-Muna, M., & Mugendi, D. (2014). Socioeconomic factors influencing levels of knowledge in soil fertility management in the Central Highlands of Kenya. *Journal of Agricultural Science and Technology*, 2014(4), 701–711.
- Mäder, P., Fliessbach, A., Dubois, D., Gunst, L., Fried, P., & Niggli, U. (2002). Soil fertility and biodiversity in organic farming. *Science*, 296(5573), 1694–1697.
- Mahmood, F., Khan, I., Ashraf, U., Shahzad, T., Hussain, S., Shahid, M., Abid, M., & Ullah, S. (2017). Effects of organic and inorganic manures on maize and their residual impact on soil physico-chemical properties. *Journal of Soil Science and Plant Nutrition*, 17(1). https://doi.org/10.4067/ S0718-95162017005000002
- Malpass, D. (2022). A transformed fertilizer market is needed in response to the food crisis in Africa. *World Bank Blogs*, December 21, 2022. https://blogs.worldbank.org/en/voices/transformed-fertilizer-market-needed-response-food-crisis-africa
- Manlay, R. J., Feller, C., & Swift, M. J. (2007). Historical evolution of soil organic matter concepts and their relationships with the fertility and sustainability of cropping systems. *Agriculture, Ecosystems & Environment, 119*(3), 217–233. https://doi.org/10.1016/j.agee.2006.07.011
- Mapanda, F., Wuta, M., Nyamangara, J., & Rees, R. M. (2011) Effects of organic and mineral fertilizer nitrogen on greenhouse gas emissions and plant-captured carbon under maize cropping in Zimbabwe. *Plant and Soil*, *343*(1-2), 67–81.
- Marenya, P. P., & Barrett, C. B. (2009a). Soil quality and fertilizer use rates among smallholder farmers in western Kenya. *Agricultural Economics*, *40*(5), 561–572. https://doi.org/10.1111/j.1574-0862.2009.00398.x
- Marenya, P. P., & Barrett, C. B. (2009b). State-conditional fertilizer yield response on western Kenyan farms. *American Journal of Agricultural Economics*, *91*(4), 991–1006. https://doi.org/10.1111/j.1467-8276.2009.01313.x
- Martey, E. (2018). Welfare effect of organic fertilizer use in Ghana. *Heliyon*, 4(10), e00844. https://doi. org/10.1016/j.heliyon.2018.e00844
- Masarirambi, M., Hlawe, M., Oseni, O., & Sibiya, T. (2010). Effects of organic fertilizers on growth, yield, quality and sensory evaluation of red lettuce (Lactuca sativa L.) 'Veneza Roxa.' *Agriculture and Biology Journal of North America*, *1*(6), 1319–1324. https://doi.org/10.5251/abjna.2010.1.6.1319.1324
- Means, M., Crews, T., & Souza, L. (2022). Annual and perennial crop composition impacts on soil carbon and nitrogen dynamics at two different depths. *Renewable Agriculture and Food Systems*, *37*(5), 437–444. https://doi.org/10.1017/S1742170522000084
- Moebius-Clune, B. N., van Es, H. M., Idowu, O. J., Schindelbeck, R. R., Kimetu, J. M., Ngoze, S., Lehmann, J., & Kinyangi, J. M. (2011). Long-term soil quality degradation along a cultivation chronosequence in western Kenya. *Agriculture, Ecosystems & Environment, 141*(1), 86–99. https://doi. org/10.1016/j.agee.2011.02.018

- Moeskops, B., Buchan, D., Van Beneden, S., Fievez, V., Sleutel, S., Gasper, M. S., D'Hose, T., & De Neve, S. (2012). The impact of exogenous organic matter on SOM contents and microbial soil quality. *Pedobiologia*, *55*(3), 175–184. https://doi.org/10.1016/j.pedobi.2012.03.001
- Muluneh, M. W., Talema, G. A., Abebe, K. B., Dejen Tsegaw, B., Kassaw, M. A., & Teka Mebrat, A. (2022). Determinants of organic fertilizers utilization among smallholder farmers in South Gondar Zone, Ethiopia. *Environmental Health Insights, 16*, 11786302221075448. https://doi. org/10.1177/11786302221075448
- Murwira, H. R. (1994). Synchrony relationships of nitrogen release and plant uptake in a Zimbabwean sandy soil amended with manure and fertiliser nitrogen. *African Crop Science Journal*, 2(1), 69–77.
- Mutuku, E. A., Roobroeck, D., Vanlauwe, B., Boeckx, P., & Cornelis, W. M. (2020). Maize production under combined Conservation Agriculture and Integrated Soil Fertility Management in the sub-humid and semi-arid regions of Kenya. *Field Crops Research*, *254*, 107833. https://doi.org/10.1016/j. fcr.2020.107833
- Mwaura, G. G., Kiboi, M. N., Mugwe, J. N., Nicolay, G., Bett, E. K., Muriuki, A., Musafiri, C. M., & Ngetich, F. K. (2021). Economic evaluation and socioeconomic drivers influencing farmers' perceptions on benefits of using organic inputs technologies in upper eastern Kenya. *Environmental Challenges*, 5, 100282. https://doi.org/10.1016/j.envc.2021.100282
- Nandwa, S. M., & Bekunda, M. A. (1998). Research on nutrient flows and balances in East and Southern Africa: State-of-the-art. *Agriculture, Ecosystems & Environment,* 71(1), 5–18. https://doi.org/10.1016/S0167-8809(98)00128-5
- Ndambi, O. A., Pelster, D. E., Owino, J. O., De Buisonjé, F., & Vellinga, T. (2019). Manure management practices and policies in sub-Saharan Africa: Implications on manure quality as a fertilizer. *Frontiers in Sustainable Food Systems*, *3*, 29. https://doi.org/10.3389/fsufs.2019.00029
- Nelson, P. N., & Su, N. (2010). Soil pH buffering capacity: A descriptive function and its application to some acidic tropical soils. *Soil Research*, *48*(3), 201–207. https://doi.org/10.1071/SR09150
- Nest, T. V., Vandecasteele, B., Ruysschaert, G., Cougnon, M., Merckx, R., & Reheul, D. (2014). Effect of organic and mineral fertilizers on soil P and C levels, crop yield and P leaching in a long term trial on a silt loam soil. *Agriculture, Ecosystems & Environment, 197*, 309–317. https://doi.org/10.1016/j. agee.2014.07.019
- Ng'ang'a, J. (2024). Kenya achieves key African target in fertilizer usage. *Kenya News Agency, April 24, 2024.* https://www.kenyanews.go.ke/kenya-achieves-key-african-target-in-fertilizer-usage/
- Nsoanya, L. N., & Nweke, I. A. (2015). Economic returns of integrated use of poultry manure and inorganic fertilizers on maize yield in Akure, Nigeria. *International Journal of Research in Agriculture and Forestry*, 2(3), 14–19.
- Nyamangara, J., Bergström, L. F., Piha, M. I., & Giller, K. E. (2003). Fertilizer use efficiency and nitrate leaching in a tropical sandy soil. *Journal of Environmental Quality*, *32*(2), 599–606. https://doi.org/10.2134/jeq2003.5990
- Nyberg, G., Bargués Tobella, A., Kinyangi, J., & Ilstedt, U. (2012). Soil property changes over a 120-yr chronosequence from forest to agriculture in western Kenya. *Hydrology and Earth System Sciences*, 16(7), 2085–2094. https://doi.org/10.5194/hess-16-2085-2012
- Odhiambo, J. J. O., & Magandini, V. N. (2008). An assessment of the use of mineral and organic fertilizers by smallholder farmers in Vhembe district, Limpopo province, South Africa. *African Journal of Agricultural Research*, *3*(5), 357–362. https://doi.org/10.5897/AJAR.9000038
- Ogunlela, V. B., Masarirambi, M. T., & Makuza, S. M. (2005). Effect of cattle manure application on pod yield and yield indices of okra (*Abelmoschus esculentus* L. Moench) in a semi-arid sub-tropical environment. *Journal of Food Agriculture and Environment*, 3(1), 125–129.
- Okuma, L., & Isiorhovoja, R. A. (2017). Farmers' perception and willingness to pay for organic fertilizer in Delta State, Nigeria. Journal of Food Agriculture and Environment, 4(1), 9–20.

- Olwande, J. (2019). Technical efficiency, soil fertility and farmers' perception: Evidence from smallholder maize production in Kenya. Sixth International Conference, September 23–26, 2019, Abuja, Nigeria. Association of Agricultural Economists. https://ideas.repec.org//p/ags/aaae19/295806.html
- Oyetunde-Usman, Z., Ogunpaimo, O. R., Olagunju, K. O., Ambali, O. I., & Ashagidigbi, W. M. (2021). Welfare impact of organic fertilizer adoption: Empirical evidence from Nigeria. *Frontiers in Sustainable Food Systems*, *5*, 691667. https://doi.org/10.3389/fsufs.2021.691667
- Palm, C. A., Gachengo, C. N., Delve, R. J., Cadisch, G., & Giller, K. E. (2001). Organic inputs for soil fertility management in tropical agroecosystems: Application of an organic resource database. *Agriculture, Ecosystems & Environment*, 83(1–2), 27–42. https://doi.org/10.1016/S0167-8809(00)00267-X
- Pfotzer, G. H., & Schüler, C. (1997). Effects of different compost amendments on soil biotic and faunal feeding activity in an organic farming system. *Biological Agriculture & Horticulture, 15*(1–4), 177–183. https://doi.org/10.1080/01448765.1997.9755192
- Place, F., Barrett, C. B., Freeman, H. A., Ramisch, J. J., & Vanlauwe, B. (2003). Prospects for integrated soil fertility management using organic and inorganic inputs: Evidence from smallholder African agricultural systems. *Food Policy*, *28*(4), 365–378. https://doi.org/10.1016/j.foodpol.2003.08.009
- Ramos, T. D. M., Jay-Russell, M. T., Millner, P. D., Baron, J. N., Stover, J., Pagliari, P., Hutchinson, M., Lilley, J., Rowley, N., Haghani, V., Aminabadi, P., Kenney, A., Hashem, F., Martínez-López, B., Bihn, E. A., Clements, D. P., Shade, J. B., Sciligo, A. R., & Pires, A. F. A. (2021). Survival and persistence of foodborne pathogens in manure-amended soils and prevalence on fresh produce in certified organic farms: A multi-regional baseline analysis. *Frontiers in Sustainable Food Systems*, *5*, 674767. https://doi.org/10.3389/fsufs.2021.674767
- Roobroeck, D., Palm, C. A., Nziguheba, G., Weil, R., & Vanlauwe, B. (2021). Assessing and understanding non-responsiveness of maize and soybean to fertilizer applications in African smallholder farms. *Agriculture, Ecosystems & Environment, 305*, 107165. https://doi.org/10.1016/j.agee.2020.107165
- Rufino, M. C., Dury, J., Tittonell, P., van Wijk, M. T., Herrero, M., Zingore, S., Mapfumo, P., & Giller, K. E. (2011). Competing use of organic resources, village-level interactions between farm types and climate variability in a communal area of NE Zimbabwe. *Agricultural Systems*, *104*(2), 175–190. https://doi.org/10.1016/j.agsy.2010.06.001
- Shakoor, A., Gan, M. Q., Yin, H. X., Yang, W., He, F., Zuo, H. F., Ma, Y. H., & Yang, S. Y. (2020). Influence of nitrogen fertilizer and straw returning on CH₄ emission from a paddy field in Chao lake basin, China. *Applied Ecology and Environmental Research*, *18*(1), 1585–1600.
- Shakoor, A., Shakoor, S., Rehman, A., Ashraf, F., Abdullah, M., Shahzad, S. M., Farooq, T. H., Ashraf, M., Manzoor, M. A., Altaf, M. M., & Altaf, M. A. (2021) Effect of animal manure, crop type, climate zone, and soil attributes on greenhouse gas emissions from agricultural soils—A global meta-analysis. *Journal of Cleaner Production, 288*, 125637.
- Sheahan, M., & Barrett, C. B. (2014). Understanding the agricultural input landscape in sub-Saharan Africa: Recent plot, household, and community-level evidence. *Policy Research Working Paper No.* 7014. World Bank Group. https://doi.org/10.1596/1813-9450-7014
- Shi, L., Ding, F., Zhu, J., Ma, R., & Guo, X. (2023). Measures for controlling gaseous emissions during composting: A review. *International Journal of Environmental Research and Public Health*, 20(4), 3587.
- Sileshi, G. W., Jama, B., Vanlauwe, B., Negassa, W., Harawa, R., Kiwia, A., & Kimani, D. (2019). Nutrient use efficiency and crop yield response to the combined application of cattle manure and inorganic fertilizer in sub-Saharan Africa. *Nutrient Cycling in Agroecosystems*, *113*(2), 181–199. https://doi.org/10.1007/s10705-019-09974-3
- Singh, R. (Ed.). (2012). Organic fertilizers: Types, production and environmental impact. Nova Science Publishers.
- Smith, P., Bustamante, M., Ahammad, H., Clark, H., Dong, H., Elsiddig, E. A., Haberl, H., Harper, R., House, J., & Jafari, M. (2014). Agriculture, forestry, and other land use (AFOLU). In O. Edenhofer et al. (Eds), *Climate change 2014: Mitigation of climate change. Contribution of Working Group III to the fifth assessment report of the Intergovernmental Panel on Climate Change* (pp. 811–922). Cambridge University Press.

- Sonsri, K., Naruse, H., & Watanabe, A. (2022). Mechanisms controlling the stabilization of soil organic matter in agricultural soils as amended with contrasting organic amendments: Insights based on physical fractionation coupled with 13C NMR spectroscopy. *Science of The Total Environment*, 825, 153853.
- Stewart, W. M., Dibb, D. W., Johnston, A. E., & Smyth, A. E. (2005). The contribution of commercial fertilizer nutrients to food production. *Agronomy Journal*, 97(1), 1–6.
- Tisdall, J. M., & Oades, J. M. (1982). Organic matter and water-stable aggregates in soils. *Journal of Soil Science*, 33(2), 141–163. https://doi.org/10.1111/j.1365-2389.1982.tb01755.x
- Tittonell, P., & Giller, K. E. (2013). When yield gaps are poverty traps: The paradigm of ecological intensification in African smallholder agriculture. *Field Crops Research*, *143*, 76–90. https://doi.org/10.1016/j. fcr.2012.10.007
- Tittonell, P., Vanlauwe, B., Corbeels, M., & Giller, K. E. (2008). Yield gaps, nutrient use efficiencies and response to fertilisers by maize across heterogeneous smallholder farms of western Kenya. *Plant and Soil*, *313*(1), 19–37. https://doi.org/10.1007/s11104-008-9676-3
- Tovihoudji, P. G., Akponikpè, P. B. I., Adjogboto, A., Djenontin, J. A., Agbossou, E. K., & Bielders, C. L. (2018). Combining hill-placed manure and mineral fertilizer enhances maize productivity and profitability in northern Benin. *Nutrient Cycling in Agroecosystems*, *110*(3), 375–393. https://doi.org/10.1007/s10705-017-9872-8
- Tully, K., Sullivan, C., Weil, R., & Sanchez, P. (2015). The state of soil degradation in sub-Saharan Africa: Baselines, trajectories, and solutions. *Sustainability*, 7(6), 6523–6552. https://doi.org/10.3390/ su7066523
- United States Environmental Protection Agency. (2024). *Composting*. Sustainable Management of Food. EPA. https://www.epa.gov/sustainable-management-food/composting
- U.S. Agency for International Development. (2023). *Agriculture, food and water security—Kenya*. U.S. Agency for International Development. https://www.usaid.gov/kenya/agriculture-food-and-water-security
- van de Ven, G., Valença, A., Marinus, W., de Jager, I., Descheemaeker, K., Hekman, W., Teklu Mellisse, B., Baijukya, F., Omari, M., & Giller, K. (2021). Living income benchmarking of rural households in low-income countries. *Food Security*, *13*, 729–749.
- van Ittersum, M. K., Cassman, K. G., Grassini, P., Wolf, J., Tittonell, P., & Hochman, Z. (2013). Yield gap analysis with local to global relevance—A review. *Field Crops Research*, *143*, 4–17. https://doi. org/10.1016/j.fcr.2012.09.009
- Vanlauwe, B. (2003). Integrated soil fertility management research at TSBF: The framework, the principles, and their application. In A. Bationo (Ed.), *Managing nutrient cycles to sustain soil fertility in sub-Saharan Africa* (pp. 25–42). Academy Science Publishers . https://hdl.handle.net/10568/55337
- Vanlauwe, B., & Giller, K. E. (2006). Popular myths around soil fertility management in sub-Saharan Africa. *Agriculture, Ecosystems & Environment, 116*(1), 34–46. https://doi.org/10.1016/j.agee.2006.03.016
- Vanlauwe, B., Kihara, J., Chivenge, P., Pypers, P., Coe, R., & Six, J. (2011). Agronomic use efficiency of N fertilizer in maize-based systems in sub-Saharan Africa within the context of integrated soil fertility management. *Plant and Soil*, 339(1–2), 35–50. https://doi.org/10.1007/s11104-010-0462-7
- Villat, J., & Nicholas, K. A. (2024). Quantifying soil carbon sequestration from regenerative agricultural practices in crops and vineyards. *Frontiers in Sustainable Food Systems*, 7. https://doi.org/10.3389/fsufs.2023.1234108
- Wainaina, P., Tongruksawattana, S., & Qaim, M. (2018). Synergies between different types of agricultural technologies in the Kenyan small farm sector. *The Journal of Development Studies*, 54(11), 1974–1990. https://doi.org/10.1080/00220388.2017.1342818
- Wang, S., Li, Y., Li, Q., Ku, X., Pan, G., Xu, Q., Wang, Y., Liu, Y., Zeng, S., Fahad, S., Liu, H., & Li, J. (2024). The after-effect of organic fertilizer varies among climate conditions in China: A meta-analysis. *Agronomy*, *14*(3), 551. https://doi.org/10.3390/agronomy14030551



- Wang, X., Yang, Y., Zhao, J., Nie, J., Zang, H., Zeng, Z., & Olesen, J. E. (2020). Yield benefits from replacing chemical fertilizers with manure under water deficient conditions of the winter wheat – summer maize system in the North China Plain. *European Journal of Agronomy*, 119, 126118. https://doi.org/10.1016/j. eja.2020.126118
- Waqas, M., Hashim, S., Humphries, U. W., Ahman, S., Noor, R., Shoid, M., Naseen, A., Hlaing, P. T., & Lin, H. A. (2023). Composting processes for agricultural waste management: A comprehensive review. *Processes*, *11*(2), 731.
- Wawire, A. W., Csorba, Á., Tóth, J. A., Michéli, E., Szalai, M., Mutuma, E., & Kovács, E. (2021). Soil fertility management among smallholder farmers in Mount Kenya East region. *Heliyon*, 7(3), e06488. https://doi.org/10.1016/j.heliyon.2021.e06488
- Wong, C., Falcone, M., Rich, G., Stubler, C., Malama, B., Lazcano, C., & Decock, C. (2023). Short-term effects of increasing compost application rates on soil C and greenhouse gas (N<sub>2</sub>O and CO<sub>2</sub>) emissions in a California central coast vineyard. *Frontiers in Environmental Science*, *11*.
- Wood, S., & Cowie, A. (2004). A review of greenhouse gas emission factors for fertiliser production. IEA Bioenergy Task 38.
- World Bank. (2024). *Fertilizer consumption (kilograms per hectare of arable land)—Kenya*. World Bank Open Data. https://data.worldbank.org/indicator/AG.CON.FERT.ZS?end=2021&locations=KE&start=1961&view=chart
- World Food Programme, & Boston Consulting Group. (2022). *Impact of increasing fertilizer prices on maize production in Kenya*. World Food Programme and Boston Consulting Group.
- Wortman, S. E., Holmes, A. A., Miernicki, E., Knoche, K., & Pittelkow, C. M. (2017). First-season crop yield response to organic soil amendments: A meta-analysis. *Agronomy Journal*, 109(4), 1210-1217.
- Wudil, A. H., Usman, M., Rosak-Szyrocka, J., Pilař, L., & Boye, M. (2022). Reversing years for global food security: A review of the food security situation in sub-Saharan Africa (SSA). *International Journal of Environmental Research and Public Health*, 19(22), 14836. https://doi.org/10.3390/ijerph192214836
- Wyer, K. E., Kelleghan, D. B., Blanes-Vidal, V., Schauberger, G., & Curran, T. P. (2022). Ammonia emissions from agriculture and their contribution to fine particulate matter: A review of implications for human health. *Journal of Environmental Management*, 323, 116285. https://doi.org/10.1016/j. jenvman.2022.116285
- Zhai, L., Wang, Z., Zhai, Y., Zhang, L., Zheng, M., Yao, H., Lv, L., Shen, H., Zhang, J., Yao, Y., & Jia, X. (2022). Partial substitution of chemical fertilizer by organic fertilizer benefits grain yield, water use efficiency, and economic return of summer maize. *Soil and Tillage Research*, *217*, 105287.
- Zhang, C., Xu, T., Feng, H., & Chen, S. (2019). Greenhouse gas emissions from landfills: A review and bibliometric analysis. *Sustainability*, *11*(8), 2282.
- Zhou, Z., Zhang, S., Jiang, N., Xiu, W., Zhao, J., & Yang, D. (2022). Effects of organic fertilizer incorporation practices on crops yield, soil quality, and soil fauna feeding activity in the wheat-maize rotation system. *Frontiers in Environmental Science*, *10*, 1058071. https://doi.org/10.3389/fenvs.2022.1058071
- Zingore, S., Murwira, H. K., Delve, R. J., & Giller, K. E. (2007). Soil type, management history and current resource allocation: Three dimensions regulating variability in crop productivity on African smallholder farms. *Field Crops Research*, *101*(3), 296–305. https://doi.org/10.1016/j.fcr.2006.12.006
- Zingore, S., Mutegi, J., Agesa, B., Tamene, L. D., & Kihara, J. (2015). Soil degradation in sub-Saharan Africa and crop production options for soil rehabilitation. *Better Crops with Plant Food*, 99(1), 24–26. https://cgspace.cgiar.org/handle/10568/68702

## Appendix A: Soil fertility and soil organic matter

#### A.1 Definition of soil fertility and soil organic matter

Soil fertility refers to a soil's ability to provide essential plant nutrients, and to maintain the necessary chemical, physical, and biological properties to sustain plant growth (Food and Agriculture Organization [FAO], 2022). Soil chemical properties are those properties, such as pH or cation exchange capacity, that influence the availability of nutrients; soil physical properties relate to soil structure, which drives the flow of gas, water, and nutrients through soil; and soil biological properties allow the continuity of nitrogen and phosphorus cycles, which provide usable forms of these nutrients for plants. Plants typically require large quantities of macronutrients, namely nitrogen (N), phosphorus (P) and potassium (K), and smaller quantities of micronutrients, such as iron (Fe), manganese, and zinc (Zn). All these elements are made available to plants through biological cycles, and chemical and physical processes, through the mineralization of soil organic matter, or through addition as inputs (Food and Agriculture Organization [FAO], 2022).

Soil organic matter (SOM) (or the closely related soil organic carbon—SOC) can be defined as organic materials found in soils; these materials originate from living organisms, primarily plants and microbes (Chenu et al., 2024). The level of SOM in soils is an essential component of soil fertility, and it contributes to crop productivity (Lal, 2006; Manlay et al., 2007). Many studies across sub-Saharan Africa and the rest of the world show the strong relationship between crop yields and SOM (Lal, 2006). SOM confers many benefits: It contains essential plant nutrients which are mineralized and become available for plant uptake (Kimetu et al., 2008); it provides cation exchange sites which are critical for retaining plant nutrients in soils (Kimetu et al., 2008); it increases the water-holding capacity of soils (Kimetu et al., 2008); it contributes to better infiltration (Franzluebbers, 2002); and it contributes to the pH buffering capacity of the soil (Nelson & Su, 2010; Adekiya et al., 2020).

SOM content is determined by a dynamic process of gains and losses of organic carbon (C). Organic material may be added to soils naturally as C assimilated by plants as they grow, and it may be brought to the field from external sources (Tittonell & Giller, 2013). On the other hand, organic material is lost from soil when crops are harvested, when other biomass is removed from fields, when crop residues are burnt or grazed, when carbon-rich topsoil is eroded, and when organic material is degraded by microbes into carbon dioxide (CO<sub>2</sub>) (Ölinger, 1996). In addition, SOM is also reduced by agricultural practices such as repeated soil disturbance or the dominance of annual crops (Means, Crews & Souza, 2002). As a result, maintaining and improving SOM is a persistent challenge for smallholder farmers (Willy et al., 2019). Furthermore, a net depletion of SOM could potentially lead to a vicious cycle where a reduction in SOM leads to soils losing productive capacity, which in turn reduces biomass production, which in turn leads to a further reduction in SOM (Kimetu et al., 2008).

#### A.2. Methods that increase soil organic matter

There are many mechanisms through which farmers can increase the SOM content of their soils and in turn improve soil fertility. Each method has its advantages and disadvantages.

SOM may be improved by traditional fallowing, by which farmers periodically leave a portion of their land unmanaged for one or more seasons, which allows vegetation to grow and improves soil fertility through lower mineralization rates, higher retention of N deposited from the atmosphere, higher uptake of N by

deeper rooting plants, higher biological N fixation, and higher input of more recalcitrant organic matter (Giller et al., 1997). However, with fallowing, gains in SOM rapidly decrease once the soil is cultivated again, and fallowing land may not be feasible for smallholders, who have limited areas of land and urgent income needs.

Intercropping with trees and other beneficial plants is another way to improve the N and C content of soil. Intercropping with grain legumes increases biological N fixation, and some grain legumes also provide direct N additions to the soil. Increased N in the soil can support subsequent C assimilation by other higher biomass crops. However, intercropping by itself is unlikely to reverse soil fertility decline without additional measures; inter-crops also compete with crops for sunlight, nutrients and water (Giller et al., 1997).

Planting N-fixing or high biomass plants, known as cover crops, can rapidly improve N content, and plant residue and C accumulation (Giller et al., 1997). However, this requires sufficient land and water, which could otherwise be used for crops themselves. Again, for smallholders with limited resources, growing a crop with no direct monetary or nutritional benefit is a challenge (Tittonell et al., 2008; Giller et al., 1997).

SOM may also be improved through better management of crop residues, which are the plant materials that are left on the field after the removal of the harvested portion. This can be achieved by reducing both the burning of crop residue and the export of crop residues from fields after the growing season, such as for livestock feed. These crop residues are an important source of organic matter which can be returned to the soil for nutrient recycling, and improved crop residue management is shown to support improvements in SOM and soil fertility. Higher levels of SOM improve the soil's physical, chemical, and biological properties (Kumar & Goh, 2000). However, on its own crop residue management cannot reverse declines caused by negative nutrient balances and related reinforcement loops that reduce biomass production. In addition, a competing use for crop residues is as livestock feed, and thus crop residues may not be available for soil fertility management (Lal, 2006).

Organic fertilizers accessible to farmers in their farmyards include animal manures, and composted or vermicomposted crop residues or farmyard waste. The main benefits of on-farm compost or manure is its low cost, nutrient cycling and waste management within the farm (Waqas et al., 2023). However, the availability of organic material or on-farm manure is often limited (Tittonell et al., 2008), and their nutrient content is highly variable.

Lastly, farmers may purchase commercial organic fertilizers, which are produced at large-scale using improved composting or vermicomposting techniques. This allows the standardization of nutrient content and control over the process. However, large-scale production could be resource intensive, and purchases of commercial organic fertilizers imply that farmers face additional costs.

## Appendix B: A case study of a specific commercial organic fertilizer

#### **B.1** Introduction

Evergrow is a commercial organic fertilizer locally manufactured by Regen Organics in Kenya. Regen Organics sold an estimated 3,600 metric tons of Evergrow to farmers in 2022, which represents a market share of more than 50 percent by quantity sold in Kenya (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b). Given its majority share of the commercial organic fertilizer market in Kenya, and the availability of information on its use and impacts from the manufacturer, we focus on Evergrow to understand adoption, impact, and benefits and challenges relating to commercial organic fertilizers. In this case study, we summarize the insights learned from three sets of surveys and three sets of farm trials that were conducted in Kenya by multiple organizations.<sup>3</sup>

Evergrow is made of residual organic materials from agricultural, animal, and municipal sources. These residual organics are composted using thermophilic aerated windrow composting. This is a common composting process that breaks down organic raw materials and eliminates pathogens. The resulting organic fertilizer contains 1.5–3 percent N, 1 percent P, 1 percent K and 20:1 C:N ratio (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b). As indicated in this paper, organic fertilizers help increase soil carbon, leading to multiple improvements in soil productivity, such as microbial activity, improved water retention, and improved nutrient holding capacity. Many studies have shown the application of organic fertilizers to improve soil fertility, while providing crops with essential nutrients. Consistent with this, Evergrow is positioned by Regen Organics as a complement to inorganic fertilizer: Evergrow may maintain yield by replacing up to 50 percent of inorganic fertilizer used on farms (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b).

Regen Organics indicates that Evergrow is designed for use on any crop, and they particularly recommend its use on horticultural crops such as tomatoes, capsicum, onions, French beans, and flowers, and high-value seasonal crops like coffee, tea, rice, and wheat. Evergrow is compliant with requirements set by the Kenya Bureau of Standards (KEBS) (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b).<sup>4</sup>

#### B.2 Farmer awareness, adoption, and perceptions of Evergrow

Strathmore Business School (SBS) and Boston Consulting Group (BCG) conducted surveys with 671 farmers and 584 agrovets (resellers of farm inputs) across six counties in Kenya (Boston Consulting Group, 2023). Farmer respondents include 74 percent and 26 percent of male and female farmers, respectively. Maize was the most commonly grown crop (68 percent of farmers), followed by beans (31 percent of farmers). Among all farmer respondents, 85 percent had heard of specific commercial

<sup>3</sup> The specific studies included are a survey with farmers and agrovets conducted by Strathmore Business School (SBS) and Boston Consulting Group (BCG), a survey with farmers conducted by 60 Decibels, a survey with farmers conducted by Kearney, a field trial conducted by Farm Star, a field trial conducted by SGS Kenya Limited, and a set of field trials conducted by Crop Nuts.

<sup>4</sup> Evergrow Gold meets EcoCert export standards (Commercial Agriculture for Smallholders and Agribusiness [CASA], 2023b).



organic fertilizers. Of those farmers who were able to name brands, 43 percent had heard of Evergrow, making it the best-known brand in this sample. While 70 percent of all farmer respondents had considered buying commercial organic fertilizer, 49 percent of farmers bought it at some point, and 43 percent of farmers bought it in the past 12 months. Evergrow was bought by 26 percent of farmers in the past 12 months, indicating that it also was the most bought brand with this sample of farmers. Of farmers who had not considered purchasing commercial organic fertilizer, 72 percent indicated that they used their own manure or compost, and 25 percent indicated that commercial organic fertilizer was unavailable in their area.

The majority of farmer respondents (94 percent) applied fertilizer in the last season. Among them, 72 percent used synthetic fertilizer, 54 percent used manure or compost, and 26 percent used commercial organic fertilizer. In terms of application rate, compared to what farmers perceived to be correct rates, farmers thought that they applied 109 percent less commercial organic fertilizer and 20 percent less synthetic fertilizer than they should, while applying 31 percent more manure or compost than what is needed. Farmers perceived the main benefits of organic fertilizers as increasing soil productivity and leading to higher crop yield, while the principal disadvantage was affordability. Similarly, agrovets considered that commercial organic fertilizers had the same yield and soil health benefits as in-house manure or compost, but at a higher cost, making it difficult to compete price-wise against in-house manure or compost. These responses imply concerns about the value for money of commercial organic fertilizers and indicate insufficient awareness of the difference between commercially produced organic fertilizer and homemade manure and composts.

Across all farmer respondents, commercial organic fertilizer had a low Net Promoter Score (NPS) of -22, comparable to synthetic fertilizer (NPS of -23), with more farmers preferring to recommend in-house manure or compost (NPS of 22). However, among Evergrow users, the product had an NPS of 36, higher than for in-house manure or compost (NPS of 21) and general commercial organic fertilizer (NPS of 23). These ratings indicate that farmers who used Evergrow saw value in it compared to their own manure or homemade compost.

In another survey, conducted by 60 Decibels, 287 Kenyan farmers who had purchased Evergrow before were interviewed (60 Decibels, 2023). This sample of respondents includes 53 and 47 percent of male and female farmers, respectively. Their median cultivation area of last year was 2 acres. 78, 43, and 26 percent of farmers selected maize, beans, and potatoes as their top 3 important crops. The median tenure of using Evergrow fertilizer was 1.5 years. Before starting to use Evergrow fertilizer, 88 percent of farmers only used chemical fertilizers; 84 percent said they had no prior access to a comparable fertilizer, and 88 percent said they could not easily find a good alternative to Evergrow. The NPS for Evergrow reported by farmer respondents was 56, which is in the top quintile of 60 Decibels' East Africa benchmark, and 82 percent of respondents consider Evergrow fertilizer's value for money to be good or very good. Nevertheless, 19 percent of respondents experienced challenges, such as high prices, poor farming outcomes, and unavailability of smaller quantities.

#### B.3 Impact of Evergrow on yields and revenues

To understand Evergrow fertilizer's impact on yield and revenue, we examine information from both self-report surveys, in which respondents shared their experiences before and after adopting Evergrow, and farm trials that rigorously measure the impact of Evergrow.

In the SBS-BCG survey mentioned above, 545 farmers reported their yields. Conducting ordinary least squares analysis with this sample, we find that adoption of commercial organic fertilizer is associated

with 31.7 percent higher yields per acre (statistically significant at p < 0.05), when controlling for the type of crop (i.e., fruits, vegetables, rice, cereals, and cash crops) and usage of other fertilizers.

In the 60 Decibels survey mentioned above, comparing their experience before and after using Evergrow fertilizer, 96 percent of farmers reported an increase in their production, and 85 percent of farmers reported an increase in their farm earnings, mostly attributed to an increase in harvest volume sold. Most farmers (90 percent) who experienced increases in total production farmed on the same amount of land, indicating that they achieved higher yields. Moreover, 83 percent of farmers reported reducing the usage of chemical fertilizer because of Evergrow fertilizer, including 42 percent who no longer used inorganic fertilizer at all, and 77 percent of farmers noticed a decrease in the occurrence of crop diseases or pests since they started using Evergrow fertilizer. Overall, 92 percent of farmers reported that their quality of life improved because of using Evergrow fertilizer, due to factors such as increased food availability at home, increased income, and improved ability to pay essential bills and education.

In another survey, conducted by Kearney, agricultural production, cost, and revenue were collected from 81 farmers who purchased Evergrow fertilizer, for both the season before they used Evergrow fertilizer and the season they used it (Kearney, 2023). Conducting ordinary least squares analysis with this sample, we find that farmers' self-reported yields per acre across multiple crops are, on average, 89 percent higher after starting to use Evergrow fertilizer (a median increase of 43 percent), and positively associated with the quantity of Evergrow that they apply per acre, although those two relationships between Evergrow usages and yields are not statistically significant. Furthermore, farmers reported their revenue increased by 217 percent on average (a median increase of 91 percent). 37 percent of this increase in revenue was attributed to Evergrow fertilizer usage, while other reasons, such as better weather conditions and increased price of harvest sales, generated the rest of the increase. The value of the annual income increase per farmer attributable to use of Evergrow Organic Fertiliser was \$1,486 and \$7,361 for small and large farmers, respectively<sup>5</sup>, equivalent to increases of 35 percent and 32 percent respectively as compared to the income reported for the season before adoption of Evergrow. Farmers reported using this increase in income primarily to fund better access to education (mentioned by 78 percent) and increased savings (mentioned by 62 percent) as well as to purchase inputs for the following season (44 percent).

Moving to farm trials, Farm Star conducted a field trial on French beans in Kenya in 2014. Specifically, it performed two treatments side-by-side, in which the control block used chemical fertilizer and the treatment block used both chemical fertilizer and Evergrow organic fertilizer. The trial results show that the yield in the treatment block with Evergrow fertilizer is more than doubled compared to the yield in the control block, 1.9 MT/ha vs. 0.9 MT/ha.

In 2023, SGS Kenya Limited conducted a farm trial on tomatoes, and Crop Nuts conducted a series of farm trials on maize, cabbage, potato, onion, and black nightshade. All those trials experimented with different combinations of chemical fertilizer and Evergrow organic fertilizer in various quantities. Those trial results show that substitution of higher rates of inorganic fertilizers with commercial organic fertilizer results in similar yields, and this finding is consistent across crops. The reduction in inorganic fertilizer application combined with small amounts of commercial organic fertilizer results in the highest benefit-cost ratio, defined as the total crop value minus fertilizers reveals a positive interaction effect of the two, indicating greater efficiency of inorganic fertilizer applications is achieved by applying commercial organic fertilizers in combination with inorganic fertilizers.

<sup>5</sup> Small farmers defined as farmers with farm size less than 10 acres, large farmers defined as farmers with farm acreage > 10 acres.

In the field trial on maize, Crop Nuts observed another advantage of Evergrow fertilizer. It reported that "as the season progressed, the site was affected by a prolonged drought which had an impact on the crop. Some plants succumbed to this drought. The plots with the Evergrow products had more plants (higher survival) and taller plants (healthier crop), suggesting that the Evergrow products can be beneficial to provide a better resistance to drought situations."

#### **B.4** Conclusion

Evergrow is a commercially available organic fertilizer with the potential to significantly positively impact the agriculture sector in sub-Saharan Africa. Internal field trials and surveys with Evergrow fertilizer users suggest positive impacts on agricultural yields and smallholder farmer incomes across multiple crops. These findings are promising, and further field trials and impact evaluations could provide more rigorous causal evidence. While farmer awareness and adoption of Evergrow fertilizer is currently low, increased availability and uptake among farmers in Kenya and other areas of sub-Saharan Africa could have large aggregate impacts on improving soil fertility, agricultural productivity, and, ultimately, smallholder farmers' livelihoods.



